Copyright is owned by the Author of the thesis. Permission is given for a copy to be downloaded by an individual for the purpose of research and private study only. The thesis may not be reproduced elsewhere without the permission of the Author.

# Telepractice with Adults with Dysphagia Post-stroke: A Survey of New Zealand Speech-Language Therapists' Perceptions

A thesis presented in partial fulfilment of the requirements for the degree of Master of Speech and Language Therapy At Massey University, Albany, New Zealand

Reshma Sreekumar

2022

#### ABSTRACT

The study explored the perceptions of speech-language therapists (SLTs) in New Zealand on the use of telepractice with adults with dysphagia post-stroke along with the perceived benefits, facilitators, barriers and beliefs. A cross-sectional, predominantly quantitative online survey design was employed via the survey platform Qualtrics. Participants were six SLTs working in New Zealand with experience using telepractice with adults with dysphagia poststroke.

There were a number of key findings. The results suggested many benefits on the use of telepractice with adults with dysphagia post-stroke: possibility to work from home, reduced travel and costs, use of a support person, and employees' better quality of life. Also, the perceptions of the SLTs with respect to the client, clinician and the workplace/organisation were elicited. In addition, the different practical issues that need consideration while working with adults with dysphagia post-stroke via telepractice were discussed. Directions for future research based on the findings of the study were also identified.

#### ACKNOWLEDGEMENTS

I, Reshma Sreekumar, at this moment take the opportunity to thank all those beautiful people who inspired and encouraged me throughout my life to achieve my goals.

I would like to acknowledge my supervisors, Dr Mershen Pillay, and Dr Sally Clendon.

To Dr Mershen Pillay, I express sincere thanks from the bottom of my heart for all your support in sharing your wisdom, competence and expertise with me throughout my study.

I thank you also for the meticulous critical suggestions and feedback on every draft and for supporting me throughout, without which I would not have been able to complete this work.

To Dr Sally Clendon, I express my sincere gratitude for all the support rendered during my difficult times which helped empower me, for your quick and complete constructive feedback and suggestions, and your unwavering support.

I would like to convey my sincere gratitude to Linda Oostenrijk and Djhoanna Lambert for introducing me to Massey University.

Thank you, Emily Jones, Senior Professional Clinician, Massey University, for letting me to be a part of your fantastic class on dysphagia, along with undergraduate students.

I would like to thank Patty Govender, Clinical Educator at the Institute of Education, Massey University, for providing excellent feedback on my draft survey questionnaire.

Thanks to all the participants for your valuable time participating in my research.

I would like to acknowledge the work done by Kristy and Mable, for writing support and the prereading team for all your efforts.

I would like to show my deep appreciation to the Massey University College of Humanities and Social Sciences for awarding me the International Excellence Scholarship, 2020.

I take this opportunity to thank the most remarkable persons in my life who are my greatest inspiration and support system. Family is an emotional word to me. I do not have enough words to express my love and gratitude towards them. Thank you Acha for being there with me to pick me up when I fall and to say that " It is ok, you can do it. We have faced so many challenges, and this shall also pass." To my Amma, thanks for being the strong pillar of support, always with

powerful words that have inspired me to go through my tough times that helped me be confident and improve myself. You have moulded me to be a good person. I am so blessed to call you both 'Acha' and 'Amma'.

My dearest Chettan, it has been years since we met but you are permanently close by my heart; no matter how hard we fought as all siblings do, in the end, we always loved each other. You have always protected and supported me secretly.

My dear Thara Chechi, thanks for coming to our family and looking after us as your own. No matter what, we will always be there for you.

I also would like to thank all my dear friends at each stage of my life for being with me throughout these days, which would turn into a memorable one. Thank you, dear ones, for endless fun, laughter, dates, trips, phone calls, holding each one's back. It's tough to find friends like you.

Thank you all!!!

## **Table of Contents**

| Abstract                   | ii  |
|----------------------------|-----|
| Acknowledgements           | iii |
| Table of contents          | v   |
| List of figures and tables | xi  |

## **CHAPTER ONE: INTRODUCTION**

| 1.1 Background               | 1 |
|------------------------------|---|
| 1.2 Rationale for this study | 2 |
| 1.3 Research question        | 3 |
| 1.4 Structure of the thesis  | 4 |

## **CHAPTER TWO: LITERATURE REVIEW**

| 2.1 Introduction                                       | 5  |
|--|----|
| 2.2 Telepractice                                       | 5  |
| 2.2.1 Definitions                                      | 5  |
| 2.2.2 Synchronous and asynchronous telepractice        | 6  |
| 2.2.3 Technology involved                              | 6  |
| 2.2.4 Benefits and outcomes of the use of telepractice | 7  |
| 2.2.5 Global perspectives                              | 8  |
| 2.2.6 COVID-19 impacts                                 | 8  |
| 2.3 Telepractice in Speech and Language Therapy        |    |
| 2.3.1 Types of clinical services                       | 9  |
| 2.3.2 Types of client groups                           | 10 |
| 2.3.3 Types of workplace settings                      | 10 |
| 2.3.4 Benefits and outcomes                            | 11 |
| 2.3.5 COVID-19 impacts                                 | 12 |
| 2.4 Telepractice for Individuals with dysphagia        | 12 |

| 2.4.1 Application of telepractice in the management of          | 12 |
|---|----|
| dysphagia   |    |
| 2.4.2 COVID-19 impacts  | 13 |
| 2.5 Benefits of clinical services for adults with dysphagia     | 13 |
| post-stroke via telepractice                                    |    |
| 2.5.1 Client-based benefits                                     | 13 |
| 2.5.2 Clinician-related benefits                                | 14 |
| 2.5.3 Workplace/organisation-related benefits                   | 15 |
| 2.6 Facilitators of clinical services for adults with dysphagia | 16 |
| post-stroke via telepractice                                    |    |
| 2.6.1 Client-based facilitators                                 | 16 |
| 2.6.2 Clinician-related facilitators in the use of telepractice | 16 |
| 2.6.3 Workplace/organisation-related facilitators in the use    | 17 |
| of telepractice   |    |
| 2.7 Barriers in the use of telepractice                         | 18 |
| 2.7.1 Client-related barriers in the use of telepractice        | 18 |
| 2.7.2 Clinician-related barriers in the use of telepractice     | 19 |
| 2.7.3 Workplace/organisation-related barriers in the use        | 20 |
| of telepractice   |    |
| 2.8 Beliefs of SLTs on the use of telepractice                  | 21 |
| 2.9 Conclusion  | 22 |
| CHAPTER THREE: METHODOLOGY                                      |    |
| 3.1 Introduction  | 24 |
| 3.2 Research aims   | 24 |
| 3.3 Research design   | 24 |
| 3.4 Participants  | 24 |
| 3.5 Survey development  | 25 |

| 3.5.1 Pilot Study          | 27 |
|----------------------------|----|
| 3.6 Data collection        | 28 |
| 3.7 Data analysis          | 28 |
| 3.8 Ethical considerations | 29 |
| 3.9 Conclusion             | 29 |

## **CHAPTER FOUR: RESULTS**

| 4.1 Introduction   | 31 |
|--|----|
| 4.2 Survey results   | 31 |
| 4.2.1 Demographics of the participants   | 31 |
| 4.2.1.1 SLTs' number of years of experience  | 31 |
| 4.2.1.2 Qualification of SLTs  | 32 |
| 4.2.1.3 Learning about dysphagia   | 32 |
| 4.2.1.4 Professional learning and development of SLTs  | 33 |
| in telepractice  |    |
| 4.2.1.5 Regions where SLTs are based in New Zealand  | 34 |
| 4.2.1.6 Employment status of SLTs  | 34 |
| 4.2.1.7 Facilities where clinical services are provided by SLTs to adults with dysphagia post-stroke | 34 |
| 4.2.2 Clinical services to adults with dysphagia post-stroke   | 35 |
| 4.2.2.1 Number of adults with dysphagia post-stroke treated by SLTs in the last two years            | 35 |
| 4.2.2.2 Age group of adults with dysphagia post-stroke seen by SLTs                                  | 36 |
| 4.2.3 Confidence with technology   | 37 |
| 4.2.3.1 Level of confidence in using technology in general   | 37 |
| 4.2.3.2 Level of confidence of SLTs in using   | 37 |
| technology in clinical practice  |    |
| 4.2.4 Telepractice service to adults with dysphagia  | 37 |
| post-stroke  |    |
| 4.2.4.1 Clinical services offered to adults  | 38 |
| with dysphagia post-stroke via telepractice  |    |

| 4.2.4.2 Platforms used for providing<br>clinical services to adults with dysphagia post-stroke | 38 |
|--|----|
| 4.2.4.3 Items used to augment clinical swallow<br>evaluation via telepractice                  | 38 |
| 4.2.4.4 Client-related factors considered when   | 38 |
| deciding to offer adults with dysphagia post-stroke  |    |
| clinical services via telepractice   |    |
| 4.2.5 Expansion of clinical services for adults with   | 39 |
| dysphagia post-stroke via telepractice   |    |
| 4.2.5.1 Plan to expand clinical services for adults with                                       | 39 |
| dysphagia post-stroke via telepractice   |    |
| 4.2.6 Benefits of clinical services for adults with dysphagia                                  | 39 |
| post-stroke via telepractice   |    |
| 4.2.6.1 Client-related benefits  | 40 |
| 4.2.6.2 Clinician-related benefits   | 42 |
| 4.2.6.3 Workplace/organisation-related benefits  | 44 |
| 4.2.7 Facilitators of clinical services for adults with dysphagia post-stroke via telepractice | 45 |
| 4.2.7.1 Client-related facilitators  | 46 |
| 4.2.7.2 Clinician-related facilitators   | 47 |
| 4.2.7.3 Workplace/organisation-related facilitators  | 49 |
| 4.2.8 Barriers of clinical services for adults with dysphagia post-stroke via telepractice     | 50 |
| 4.2.8.1 Client-related barriers  | 51 |
| 4.2.8.2 Clinician-related barriers   | 52 |
| 4.2.8.3 Workplace/organisation-related barriers  | 53 |
| 4.3 Beliefs about telepractice   | 54 |
| 4.3.1 Level of agreement of SLTs   | 55 |
| 4.3.2 Conclusion   | 56 |

# **CHAPTER FIVE: DISCUSSION**

| 5.1 Introduction  | 57 |
|---|----|
| 5.2 Clinical services offered to adults with dysphagia              | 57 |
| post-stroke via telepractice  |    |
| 5.2.1 Telepractice provides SLTs with an option to expand           | 58 |
| services to un-served populations                                   |    |
| 5.2.2 Methods of augmentation of clinical practice                  | 59 |
| 5.2.3 Client-related factors considered when deciding to offer      | 60 |
| clinical services   |    |
| 5.3 Telepractice is influenced by the training and                  | 61 |
| professional development of SLTs                                    |    |
| 5.3.1 Advantages of formal training and limitations                 | 62 |
| of the curriculum in New Zealand                                    |    |
| 5.3.2 Level of confidence of SLTs                                   | 63 |
| 5. 4 Benefits and facilitators of telepractice under                | 63 |
| New Zealand context   |    |
| 5.4.1 Telepractice is an appropriate method to deliver              | 63 |
| clinical services with adults with dysphagia post-stroke            |    |
| 5.4.2 Perceived benefits of the use of telepractice in delivering   | 64 |
| clinical services   |    |
| 5.4.3 Perceived facilitators in the use of telepractice in          | 65 |
| delivering clinical services with adults with dysphagia post-stroke |    |
| 5.5 Barriers limiting the use of telepractice in New Zealand        | 67 |
| 5.6 Beliefs of SLTs in the use of telepractice with adults          | 70 |
| with dysphagia post-stroke  |    |
| 5.6.1 Telepractice is an appropriate method to deliver              | 70 |
| clinical services   |    |

| 5.6.2 Access to telepractice by adults with dysphagia post-stroke | 72 |
|---|----|
| 5.6.3 Client-clinician rapport via telepractice                   | 73 |
| 5.6.4 Workplace set up and access to telepractice                 | 73 |
| 5.6.5 Telepractice vs face-to-face clinical service delivery      | 74 |

## **CHAPTER SIX: CONCLUSION**

| 6.1 Purpose, design and rationale                         | 76 |
|---|----|
| 6.2 Limitations of the study                              | 77 |
| 6.3 Strengths of the study                                | 79 |
| 6.4 Validity and reliability of the study                 | 80 |
| 6.5 Implications for clinical practice, client education, | 80 |
| and training  |    |
| 6.6 Directions for future research                        | 81 |
| 6.7 Conclusion  | 82 |

## REFERENCES

## APPENDICES

| Appendix A: Recruitment Email      | 90  |
|------------------------------------|-----|
| Appendix B: Information Sheet      | 91  |
| Appendix C: Copy of Online Survey  | 93  |
| Appendix D: Ethics Approval Letter | 104 |

84

# List of Figures and Tables

# List of Figures

| Figure 1: | Number of Years Working as SLT  | 32 |
|-----------|---|----|
| Figure 2: | Events of Professional Learning and Development of the                                  |    |
|           | Participants in Telepractice  | 33 |
| Figure 3: | Facilities where Clinical Services are Provided<br>To adults with dysphagia post-stroke | 35 |
| Figure 4: | The Number of Adults with Dysphagia post-stroke   | 37 |
|           | Seen by SLTs  |    |
| List of T | ables   |    |
| Table 1:  | Different Clinical Service types delivered via Telepractice                             | 9  |
| Table 2:  | Workplace Settings of Telepractice  | 10 |
| Table 3:  | Client-based Benefits of Telepractice   | 14 |
| Table 4:  | Clinician-related Benefits of Telepractice  | 15 |
| Table 5:  | Workplace/organisation-related Benefits of Telepractice                                 | 15 |
| Table 6:  | Client-related Facilitators of Telepractice   | 17 |
| Table 7:  | Clinician-related Facilitators in the use of Telepractice                               | 18 |
| Table 8:  | Workplace-related Facilitators in the use of Telepractice                               | 19 |
| Table 9:  | Client-related Barriers in the use of Telepractice                                      | 20 |
| Table 10: | Clinician-related Barriers in the use of Telepractice                                   | 21 |
| Table 11: | Organisation-related Barriers in the use of Telepractice                                | 36 |
| Table 12: | Number of Adults with Dysphagia post-stroke treated by SLTs during the last two years   | 39 |
| Table 13: | Themes Identified under the Benefits of Telepractice                                    | 40 |

| Table 14: | Code Summary for Client-related Benefits                              | 43 |
|-----------|---|----|
| Table 15: | Code Summary for Workplace-related Benefits                           | 44 |
| Table 16: | Themes Identified under the Facilitators of Telepractice              | 45 |
| Table 17: | Code Summary for Client-related Facilitators                          | 46 |
| Table 18: | Code Summary for Clinician-related Facilitators                       | 48 |
| Table 19: | Code Summary for Workplace-related Facilitators                       | 49 |
| Table 20: | Themes Identified under the Barriers of Telepractice                  | 50 |
| Table 21: | Code Summary for Client-related Barriers                              | 51 |
| Table 22: | Code Summary for Clinician-related Barriers                           | 52 |
| Table 23: | Code Summary for Workplace-related Barriers                           | 53 |
| Table 24: | Frequency Counts of the Beliefs of SLTs about the use of Telepractice | 54 |

#### **Chapter One: Introduction**

#### 1.1 Background

The present study focuses on understanding the perceptions of speech-language therapists (SLTs) in New Zealand on the use of telepractice with adults with dysphagia post-stroke. This includes perceptions of the benefits, facilitators, barriers and beliefs. The findings will contribute to the growing body of information on telepractice in dysphagia practice, including diagnosis and intervention.

Telepractice is the use of telecommunications technology to deliver speech and language therapy (SLT) services to clients in a different location than the therapist (Cason & Cohn, 2014). It includes voice calls, video-conferencing or tele-conferencing to assess and provide therapeutic support to clients, enabling them and the clinician to meet regardless of being in different locations (Akemoglu et al., 2020). Telepractice is a viable service delivery option, particularly to meet the needs of the under-served population (Fong et al., 2020). The concept of under-served area is a region in which there is insufficient health workforce to meet population needs and also regions in which people experience barriers to health care accessibility (Wylie et al., 2013). In addition, the use of telepractice has improved clinical outcomes, reduced the costs involved, and alleviated the barriers of distance and travel (Coyle, 2012).

Telepractice is an effective service delivery option across various disorders, including dysphagia (Morrell et al., 2017). Dysphagia is any disruption in the swallowing process and can severely affect individuals' quality of life, resulting in distress in the client and their family members. In New Zealand, stroke is considered a significant cause of chronic disability and the third most common cause of death (Bay et al., 2015). However, reports on the use of telepractice

with adults with dysphagia post-stroke are almost absent in New Zealand, and hence, the need for this study.

#### 1.2 Rationale for this study

People worldwide are going through a catastrophic, life-threatening pandemic; Coronavirus disease of 2019 (COVID-19) is affecting people of all ages. Older people and those with underlying medical problems are more likely to develop severe illnesses. Given the protection levels, it is difficult for people to receive health care services (Bedford et al., 2020) including adults with dysphagia post-stroke (Maggio et al., 2020). Telepractice is convenient in such a situation and can provide health care services.

New Zealand implemented a strict lockdown on 25 March 2020, even when no deaths were reported (Klein, 2020); then, there were multiple lockdowns in different phases. These lockdowns encompassed strict home quarantine with residents to leave home only for essential purposes such as purchasing food items and medical appointments. In addition, the nation's borders were closed for non-nationals. Such lockdown measures eliminated the virus in New Zealand, almost 11 weeks after the quarantine started. As a result, from January 2020 to August 2021, there were only 2,558 confirmed cases of COVID-19, with 26 deaths in New Zealand (WHO, 2021).

Telepractice has emerged as a driving force in health care in the majority of the countries during the COVID-19 pandemic, wherein health care can be performed remotely without the person having to come to the hospital (Faux et al., 2020). When compared to other countries, the lockdown periods were short in New Zealand (during the study period). This meant that the use of telepractice may not have been as widespread as in other countries. The present study is the first in New Zealand that explores the perceptions of SLTs on the use of telepractice with adults with dysphagia post-stroke.

Evidence supports that telepractice is a viable and reliable means for screening and assessing people with various health-related needs (Kantarcigil & Malandraki, 2017; Mohan et al., 2017; Morrell et al., 2017). Telepractice also allows consultations between experts and the SLTs, thereby improving client outcomes and, considered to be cost-effective by reducing travel costs for the client and the SLT (Burns et al., 2017; Coyle, 2012; Nordio et al., 2018). Multiple studies have focused on the efficacy of dysphagia management via telepractice (Sharma et al., 2011).

The study will add to the literature by contributing to the growing evidence base of the use of telepractice with adults with dysphagia post-stroke. It will also provide critical information about the New Zealand context. Understanding the perceptions of SLTs working with adults with dysphagia post-stroke in New Zealand will inform the use of telepractice. In addition, this will also help to identify possible solutions to overcoming barriers in using this technology.

#### **1.3 Research question**

The study aims to answer the following research question:

What do SLTs in New Zealand perceive about the use of telepractice with adults with dysphagia post-stroke along with the factors they perceive as barriers, beliefs and facilitators?

#### **1.4 Structure of the thesis**

Chapter One of the thesis introduces the background of this study, the rationale, the research question, and the thesis' structure. Chapter Two provides a review of the relevant literature. Chapter Three outlines the methodology used including the research aims, the research design, participant recruitment, procedures, data collection and analysis and ethical considerations. Chapter Four describes the results of the study. Chapter Five discusses the study's findings compared to the relevant literature reviewed in Chapter Two. Finally, Chapter Six presents the conclusions and discusses the limitations, implications for clinical practice, client education, training, and directions for future research.

#### **Chapter Two: Literature Review**

#### **2.1 Introduction**

This chapter reviews the literature connected with the perceptions of SLTs on the use of telepractice with adults with dysphagia post-stroke. The review starts with definitions of telepractice and related terms, the types of telepractice, and the technology. Additionally, telepractice is reviewed for its benefits and outcomes, and global perspectives in general and with speech and language therapy (SLT), including its use with individuals with dysphagia post-stroke. The impacts of COVID-19 on the use of telepractice are reviewed alongside its benefits, facilitators, and barriers concerning the client, clinician and workplace. Finally, SLTs' beliefs on the use of telepractice with adults with dysphagia post-stroke are examined

#### **2.2 Telepractice**

This section reviews the definitions of telepractice, synchronous and asynchronous types of telepractice, technology involved, benefits and outcomes, and global perspectives and describes the impact of COVID-19 on the use of telepractice.

#### **2.2.1 Definitions**

According to WHO (2018), telepractice involves telecommunications and virtual technology to provide health care outside of traditional health care facilities. The American Speech-Language-Hearing Association (ASHA, 2019) provided the following definition: "Telepractice is the application of telecommunications technology to the delivery of speech-language pathology (SLP) and audiology professional services at a distance by linking clinician to clinician to clinician for assessment, intervention and /or consultation."

It is worth investigating the other terms connected with telepractice, and those used synonymously, such as telerehabilitation, telemedicine, and tele-dysphagia intervention. For example, telerehabilitation delivers services namely assessment, monitoring, intervention, supervision, education, consultation, and counselling services via information technology (Brennan et al., 2010; Richmond et al., 2017; Rogante et al., 2010; Russell, 2007).

Telemedicine is defined as the use of diagnostic, and treatment services delivered over a distance by a clinician, nurse or related professional (Fong et al., 2011). In addition, the term tele-dysphagia has emerged due to merging the terms telepractice and dysphagia (Coyle, 2012). Tele-dysphagia intervention delivers dysphagia therapy services through telecommunications technology (Cassel, 2016).

In this thesis, the term telepractice is used, even though there are different terms used to deliver dysphagia services through telecommunications channel. The term telepractice is the one that is commonly used in clinical service delivery.

#### 2.2.2 Synchronous and asynchronous telepractice

This section reviews the different types of telepractice, usually classified into synchronous (real-time interaction technology) and asynchronous (store-and-forward) types (Cherney et al., 2011; Keck & Doarn, 2014). In a real-time interaction system, services are provided immediately to individuals and can be done in a nursing home, the client's home via videoconferencing system or using telephone calls (Sasikala et al., 2018). In the "store-andforward" system, the client and the service provider need not be online simultaneously. Here, images are sent to a remote specialist. A hybrid model that combines synchronous and asynchronous technology has also been used (Cherney et al., 2011; Keck & Doarn, 2014).

#### 2.2.3 Technology involved

This section discusses the various processes involved in telepractice by using advanced technology and systems along with other appliances. Clinicians offering services via telepractice

depend upon different technological processes and solutions. The most effective tool for delivering services through telepractice is videoconferencing (Lowman & Kleinert, 2017; Sharma et al., 2011; Ward et al., 2017). Services for adults with stroke with real-time videoconferencing have achieved equivalent outcomes to a face-to-face assessment (Gregory et al., 2011).

The outcomes of telepractice are enhanced with the use of high quality videoconferencing systems and the use of peripherals. Telepractice research benefitted from more advanced videoconferencing systems capable of transmitting improved audio and video signals (Nordio et al., 2018; Sharma et al., 2011; Tinelli et al., 2017; Ward et al., 2012). The reliability of asynchronous teleconsultation and tele-assessment for adults with dysphagia was studied with three SLPs analysing the "store-and-forward" videofluoroscopic swallow assessments with a good inter-rater agreement (Malandraki et al., 2013). Combining software applications and peripheral devices with videoconferencing equipment is significant in providing health care. Pulse oximeter, medical camera, web cameras, extra lighting and mobile devices were the peripherals used to improve the information collected on dysphagia management (Grogan-Johnson et al., 2010; Ward et al., 2009).

#### 2.2.4 Benefits and outcomes of the use of telepractice

This section will review the benefits and outcomes of the use of telepractice. Telepractice allows consultations between experts and the clinician working with the client, improving treatment outcomes and cost effectiveness by reducing travel costs for the client and clinician (Cason & Cohn, 2014; Coyle, 2012; Ward et al., 2017). The authors also stated that telepractice helps to provide service to multicultural and bilingual population. As such, the need for

telepractice will grow due to four fundamental benefits: improved access, cost efficiency, better service quality and client demand (Grillo, 2019).

Assessment conducted via telepractice provides valid and reliable results compared to the clinical decisions made in face-to-face assessment (Nordio et al., 2018; Sharma et al., 2011; Tinelli et al., 2017; Ward et al., 2012). Also, telepractice enables clinicians to cover a large geographical area and provide more clinical services to clients (Mashima & Doarn, 2008). In addition, the use of telepractice also helps to provide individualised assessment and treatment in familiar surroundings, achieve better satisfaction and reduce the service costs (Burns et al., 2017; Weinstein et al., 2014).

#### **2.2.5 Global perspectives**

This section reviews the use of telepractice in the Majority World contexts. Majority World refers to countries which comprise the majority of the world's nations and who are previously termed less developed countries (LCDs) (Wylie et al., 2013). The use of telepractice in delivering health care services in Majority World contexts is on the increase (Combi et al, 2016). The growth of information and communication technology (ICT) has opened up fresh ideas around providing quality health care to people globally via telepractice (Combi et al., 2016). The governments in Majority World contexts are carrying out efforts to incorporate telepractice in their health care systems (Srivastava et al., 2021), and the COVID-19 pandemic has provided an opportunity for the improved use of telepractice in the health sector in many low and middle-income countries (LMICs).

#### 2.2.6 COVID-19 impacts

COVID-19 was declared a global pandemic (WHO, 2021). This pandemic has caused extensive devastation worldwide, leading to high mortality rates (Bedford et al., 2020). As a

result, many health care professionals faced challenges in providing health care to people through in-person care.

Telepractice is a valuable tool for avoiding direct contact between clients and clinicians, especially during the COVID-19 pandemic, when social distancing is a protocol (Faux et al., 2020; Maggio et al., 2020). As a result, telepractice has become an effective means of providing appropriate health care for people with COVID-19 and free of COVID-19 infection in different countries (Aggarwal et al., 2020; Fong et al., 2020; Rech et al., 2020).

#### 2.3 Telepractice in Speech and Language Therapy

This section reviews the use of telepractice by SLTs in various forms of health care provision in SLT.

#### 2.3.1 Types of clinical services

The different types of clinical services delivered in SLT via telepractice vary as illustrated in Table 1. This shows telepractice services used for multiple service delivery options such as treatment, consultation, screening, assessment, follow-up sessions, case review, referral purposes, teacher and family support, and management of different disorders in many countries.

#### Table 1

Different Clinical Service types delivered via Telepractice

| Clinical services  | Country          | Authors                |
|--------------------|------------------|------------------------|
| Screening          |                  |                        |
| Consultation       | United States of | Tucker (2012)          |
| Treatment          | America          |                        |
| Assessment         |                  |                        |
| Consultation       |                  |                        |
| Family support     | Australia        | Hill and Miller (2012) |
| Teacher support    |                  |                        |
| Treatment          |                  |                        |
| Follow-up sessions |                  |                        |
| Assessment         |                  |                        |
| Consultation       |                  |                        |

| Treatment                         |             |                     |
|-----------------------------------|-------------|---------------------|
| Management of dysphagia in adults | India       | Mohan et al. (2017) |
| Management of aphasia in children |             |                     |
| Management of speech disorders    |             |                     |
| Assessment                        |             |                     |
| Screening                         | Hong Kong   | Fong et al. (2020)  |
| Consultation                      |             |                     |
| Treatment                         |             |                     |
| Assessment                        |             |                     |
| Diagnosis                         | New Zealand | NZSTA (2012)        |
| Case review                       |             |                     |
| Referral purpose                  |             |                     |

The above table shows that telepractice services can be used for various service delivery options such as assessment, consultation, treatment, follow-up sessions, and referral purposes in different countries.

## 2.3.2 Types of client groups

This section presents the use of telepractice among various client groups. Telepractice is considered an effective service delivery option across a variety of disorders such as fluency disorders (Carey et al., 2010); childhood speech and language disorders (Grogan-Johnson et al., 2010; Waite et al., 2010); motor disorders in stroke (Piron et al., 2008); and dysphagia (Miles et al., 2020; Morrell et al., 2017; Soldatova et al., 2020).

## 2.3.3 Types of workplace settings

The different workplace settings in which telepractice services are provided by SLTs in two countries are depicted in Table 2.

### Table 2

Workplace Settings of Telepractice

| Telepractice settings  | Country | Authors                |
|------------------------|---------|------------------------|
| Public health facility |         | Hill and Miller (2012) |

| Private practice                 |           |                     |
|----------------------------------|-----------|---------------------|
| Public education settings        | Australia |                     |
| Community service                |           |                     |
| Specialist service               |           |                     |
| Private education settings       |           |                     |
| Self employed                    |           |                     |
| Public/private non-profit        |           |                     |
| Organisations                    | India     | Mohan et al. (2017) |
| Central-government organisations |           |                     |

Telepractice service is provided under more natural settings in-home than in a clinic (Reynolds et al., 2009). This is one of the benefits of the use of telepractice, wherein the clinical service is provided under the settings familiar to the clients. Moreover, telepractice by SLTs is not only confined to service provision in rural areas Mashima and Doarn (2008). The shortage of SLTs in many regions could be easily made up by the use of telepractice.

#### **2.3.4 Benefits and outcomes**

This section reviews the benefits and outcomes of SLT services delivered via telepractice in various forms of health care provision. The use of telepractice in SLT suggests multiple benefits and results. For example, in Australia, SLT services delivered via synchronous telepractice were more cost-efficient than standard care, with the benefits of improved service efficiency and high client and clinician satisfaction (Ward et al., 2017).

Sometimes clients cannot access SLT services locally and telepractice enables clients to access experts remotely. Other times there may be SLT services available locally, but the SLT does not have the necessary expertise. Telepractice allows the primary SLT to work with an expert SLT remotely and this can have benefits: (1) higher quality of care for the client, (2) advanced training to the primary clinician, (3) provide clinical service to multicultural and multilingual population (Cason & Cohn, 2014; Coyle, 2012; Ward et al., 2017). The other benefits include service on par with face-to-face therapy (Kantarcigil & Malandraki, 2017;

Morrell et al., 2017; Nordio et al., 2018) and that dysphagia expertise could be obtained in underserved areas (Malandraki et al., 2011). It is also found that telepractice could come in use in COVID-19 lockdown situations (Kantarcigil & Malandraki, 2017; Morrell et al., 2017; Nordio et al., 2018).

#### 2.3.5 COVID-19 impacts

COVID-19 impacts included physical distancing, stay-at-home, facemask use, and a lot of strain on health care workers worldwide. Many SLTs faced limitations in offering face-to-face health care to people during this pandemic situation, although this is to avoid direct person-toperson contact. It envisages utilising telepractice and suggests that telepractice has now emerged as a new driving force in SLT worldwide.

#### 2.4 Telepractice for Individuals with Dysphagia

This section reviews the application of telepractice in managing dysphagia and COVID-19 impacts.

#### 2.4.1 Application of telepractice in the management of dysphagia

Telepractice services are used in dysphagia management by SLTs. Research studies in The United States of America, Italy and Australia have demonstrated that telepractice is a viable and reliable means of service delivery in the screening, assessment and treatment of adults with dysphagia post-stroke leading to improvement in survival and reduction in the medical care expenditure (Kantarcigil & Malandraki, 2017; Morrell et al., 2017; Nordio et al., 2018). The researchers in these studies concluded that assessment conducted via telepractice provided valid and reliable results compared to the clinical decisions made in the face-to-face assessment. In the case of instrumental and non-instrumental dysphagia evaluation also, there has been high agreement between telepractice and face-to-face delivery of service (Burns et al., 2016). Only in cases where the severity of dysphagia was more there have been lower levels of consensus among clinicians (Morrell et al., 2017).

#### 2.4.2 COVID-19 impacts

There is potential to keep adults with dysphagia well managed at home and out of the hospital under the COVID-19 pandemic using telepractice. Telepractice is considered a favourable health care option for better-quality health care (Kantarcigil & Malandraki, 2017; Morrell et al., 2017; Weinstein et al., 2014). Furthermore, telepractice helps avoid COVID-19 virus exposure due to proximity to the aero-digestive tract while performing swallow assessments and interventions, as containment of virus spread is a primary focus in health care settings (Maggio et al., 2020; Miles et al., 2020). In addition, many dysphagia treatment techniques are classified as aerosol-generating procedures, placing health workers at risk. Thus, telepractice has become necessary in health care provision during the COVID-19 pandemic, whereby assessment and rehabilitation can be done remotely, without hospital admission (Faux et al., 2020; Miles et al., 2021; Soldatova et al., 2020). Under such circumstances, many organisations have started advocating for the use of telepractice.

#### 2.5 Benefits of clinical services for adults with dysphagia post-stroke via telepractice

This section reviews the benefits of telepractice for delivering clinical services with respect to the client, clinician and the organisation.

#### 2.5.1 Client-based benefits

The client-based benefits of clinical services for adults with dysphagia post-stroke via telepractice are listed in Table 3. Better access to dysphagia experts could significantly benefit individuals with dysphagia. Swales et al. (2020) reported the benefits of telepractice to be (i)

reduced travel time, (ii) flexibility and convenience, (iii) greater access to different models of

care, and (iv) availability of clinical services at client's home.

## Table 3

Client-based Benefits of Benefits of Clinical Services for Adults with Dysphagia post-stroke via

## Telepractice

| Benefits of telepractice                                   | Authors                       |
|--|-------------------------------|
| Better access to speech-language therapy services for an   |                               |
| underserved population                                     | Houn and Trottier (2003)      |
| Improved access to services for people with cultural and   | Mashima and Doarn (2008)      |
| linguistic diversity                                       |                               |
| Individual therapy programme                               |                               |
| Increased collaboration among team members                 |                               |
| Savings in time and costs                                  |                               |
| Improved quality of service in the client's functional     | McCue et al. (2010)           |
| environment  |                               |
| Improved access to client-centered health care             |                               |
| Reduced patient's driving time                             | Regina Molini-Avejonas et al. |
| Making health care services more accessible to individuals | (2015)                        |
| living in remote areas                                     |                               |
| High level of client satisfaction                          | Orlando et al. (2019)         |
| Providing personalised contact with families               | Wrape and McGinn (2019)       |

The major client-related benefits of clinical service for adults with dysphagia post-stroke include better access of SLT for underserved population, savings in time and costs, improved quality of service in client's functional environment.

## 2.5.2 Clinician-related benefits

This section reviews the benefits of telepractice for clinicians. Telepractice enables clinicians to cover a large geographical area and provide more clinical services to clients; this is one of the benefits. Table 4 provides the significant clinician-related benefits of telepractice with adults with dysphagia post-stroke.

Clinician-related Benefits of Telepractice with Adults with Dysphagia post-stroke.

| Clinician-related benefits of telepractice                  | Authors                  |
|---|--------------------------|
| Treatment in non-clinic settings and treating house-bound   | Mashima and Doarn (2008) |
| clients   | McCue et al. (2010)      |
| Better access to clients, better time efficiency and cost   |                          |
| efficiency  | Coyle (2012)             |
| Plan for adjusting to different factors such as cognitive   |                          |
| problems  |                          |
| Cost-effective clinical service delivery option             | Hill and Miller (2012)   |
| The quality of service offered via telepractice had similar | Lowe et al. (2013)       |
| outcomes when compared with face-to-face therapy in many    | Mashima and Doarn (2008) |
| disorders such as fluency disorders, voice disorders,       | Wales et al. (2017)      |
| dysphagia and child speech and language disorders           |                          |
| Telepractice is feasible even under COVID-19 pandemic       | Malandraki et al. (2021) |
| situation   |                          |

Better access to clients, better time and cost efficiency and the feasibility of using telepractice

under COVID-19 lockdown situations are the other benefits.

## 2.5.3 Workplace/organisation-related benefits

This section reviews the benefits of telepractice with respect to workplace/organisation and these are tabulated in Table 5

Various workplace/organisation-related benefits influence the delivery of clinical services

via telepractice. The major ones include good internet connectivity and accessories, the role of

facilitators, client candidacy criteria, privacy and safety safeguards and access to secure service.

## Table 5

Workplace/organisation-related Benefits of Telepractice with Adults with Dysphagia post-stroke

| Workplace/organisation-related benefits              | Authors                          |
|--|----------------------------------|
| Good system requirements with appropriate band width | Ayanikalath (2017); Burns et al. |
| Availability of computers and other accessories      | (2017); Malandraki et al. (2011) |
|  | Ward et al. (2013)               |
| Privacy and legal safeguards                         |                                  |
| Safety precautions                                   | Gough et al. (2015)              |

| Client candidacy criteria                                | Richmond et al. (2017) |
|--|------------------------|
| Facilitator's role                                       |                        |
| Availability of information technology support for       |                        |
| offering telepractice                                    | Pfitzner et al. (2020) |
| Organisational access to appropriate technology and good |                        |
| organisation set up                                      |                        |

#### 2.6 Facilitators of clinical services for adults with dysphagia post-stroke via telepractice

This section describes the factors that serve as facilitators in the use of telepractice with respect to the clients, clinicians and the workplace/organisation.

## 2.6.1 Client-related facilitators of clinical services for adults with dysphagia post-stroke via

#### telepractice

The important client-related facilitators in the use of telepractice include factors pertaining to the suitability of clients Ward et al. (2013). The role of facilitators is important in addressing safety issues in case of emergency and also to help with completing tasks to facilitate the assessment and treatment Malandraki et al. (2021); (Sharma et al., 2011).

# **2.6.2** Clinician-related facilitators in the use of telepractice with adults with dysphagia poststroke

The clinician-related facilitators in the use of telepractice are underlined in this section and are outlined in Table 6.

Formal training in telepractice and professional development courses and ethical guidelines are also considered facilitators by Hill and Miller (2012). The significance of training and development is mainly because telepractice involves a change in practice (Edirippulige & Armfield, 2017).

Clinician-related Facilitators in the use of Telepractice with Adults with Dysphagia post-stroke

| Clinician-related facilitators                     | Authors                          |  |
|--|----------------------------------|--|
| Policy statements for people with complex problems | Brennan et al. (2010)            |  |
| Professional development courses                   |                                  |  |
| Demonstrations                                     |                                  |  |
| Electronic assessment and treatment resources      | Hill and Miller (2012)           |  |
| Funding to establish telepractice                  |                                  |  |
| Formal training and ethical guidance               |                                  |  |
| Prior training for assessing complex cases of      | Ward at al. $(2012)$             |  |
| dysphagia  |                                  |  |
| Formal education and training for clinicians       | Edirippulige and Armfield (2017) |  |
| Need for clinical training in telepractice service | Miles et al. (2021)              |  |
| delivery   |                                  |  |

The authors also stated that the terminologies connected with telepractice, the design and clinical implications, technology applications, and national policies associated with telepractice could form the subject matter for such training.

# 2.6.3 Workplace/organisation-related facilitators in the use of telepractice with adults with dysphagia post-stroke

This section reviews the workplace/organisation-related facilitators in the use of

telepractice. The primary workplace/organisation-related facilitators are provided in Table 7.

Table 7 delineates the several workplace/organisation-related facilitators in the use of

telepractice by SLTs, which include safety precautions, legal standards, space availability in the

client's home and clinician facility and also points to the fact that the service platforms should be

secure.

Workplace/organisation-related Facilitators in the use of Telepractice with Adults with

| Workplace/organisation-related facilitators                | Authors                  |
|--|--------------------------|
| Safety precautions   |                          |
| Privacy  | Gough et al. (2015)      |
| Legal standards  | Richmond et al. (2017)   |
| Internet specifications for dysphagia assessment           |                          |
| Service delivery platforms to be secure in terms of client | Doss et al. (2018)       |
| confidentiality and data storage                           |                          |
| Appropriate space availability at the client's home and    | Burgoyne and Cohn (2020) |
| clinician facility   |                          |

To accommodate different levels of physical and psychological attributes of individuals with co-

morbidities, the service delivery system via telepractice should be flexible.

## 2.7 Barriers in the use of telepractice

This section discusses some of the significant barriers in using telepractice concerning client,

clinician and workplace/organisation.

## 2.7.1 Client-related barriers in the use of telepractice

This section will review client-related barriers in using telepractice with adults with dysphagia

post-stroke. These are presented in Table 8.

The participants perceived several client-related barriers in using telepractice in clinical service delivery. The client-related barriers considered essential in delivering clinical services include client suitability, client safety (client safety means safety measures that prevents incidents that can cause potential harm to patients from individual and organisational factors) movement and hearing disorders, and other comorbidities.

Client-related Barriers in the use of Telepractice with Adults with Dysphagia post-stroke

| Client-related barriers in the use of telepractice       | Authors                    |
|--|----------------------------|
| Case-to-case basis client appropriateness                | Brennan et al. (2010)      |
| Client factors like hearing impairment and movement      | Ward et al. (2012)         |
| disorders  |                            |
| Medical diagnosis of the clients and associated co-      | Ward et al. (2012)         |
| morbidities  |                            |
| Information provided by the clients could not be checked |                            |
| for accuracy   | Kantarcigil and Malandraki |
| Clients not remembering the names of the drugs they      | (2017)                     |
| consume  |                            |
| Client safety, as far as the management of dysphagia is  | Miles et al. (2021)        |
| concerned  |                            |
| Reimbursement hurdles                                    | Malandraki et al. (2021)   |

Reimbursement hurdles are also considered to be an important client-related barrier in the use of

telepractice with adults with dysphagia post-stroke.

## 2.7.2 Clinician-related barriers in the use of telepractice

This section reviews clinician-related barriers to telepractice. The significant clinician-related

barriers in the use of telepractice are tabulated in Table 9.

One example of a barrier was a lack of specially trained SLTs in the use of telepractice in

dysphagia assessment as instrumental swallowing assessments such as fibreoptic endoscopic

evaluation of swallowing (FEES) and modified barium swallow (MBS) have to be administered

and interpreted (Malandraki et al., 2011).

Clinician-related Barriers in the use of Telepractice with Adults with dysphagia post-stroke

| Clinician related harring  | Anthong                  |
|--|--------------------------|
| Clinician-related barriers   | Authors                  |
| Lack of skilled and specially trained staff (it is not possible to | Bours et al. (2009)      |
| have telepractice service in all settings)                         |                          |
| Lack of specially trained SLTs                                     | Malandraki et al. (2011) |
|  | May and Erickson (2014)  |
| Lack of training of SLPs in telepractice and the lack of           | Hill and Miller (2012)   |
| evidence for instrumental assessment via telepractice              |                          |
| Lack of adequate training of SLTs                                  | Hill and Miller (2012)   |
| The attitudes and perceptions of the clinicians                    | May and Erickson (2014)  |
| Difficulty in accurately assessing clients                         |                          |
|  | Pfitzner et al. (2020)   |
| Difficult to convey empathy via telepractice, when working         |                          |
| with clients with linguistically or culturally diverse             | Pfitzner et al. (2020)   |
| backgrounds  |                          |
| Lack of proximity with the client                                  | Miles et al. (2021)      |
| Reduced efficiency when clinical service is provided remotely      | Miles et al. (2021)      |
| Consent form for provision of telepractice in dysphagia            | Malandraki et al. (2021) |

### 2.7.3 Workplace/organisation-related barriers in the use of telepractice

This section reviews the workplace/organisation-related barriers on the use of telepractice. The primary workplace/organisation-related barriers are provided in Table 10. There are many workplace/organisation-related barriers in the use of telepractice. These include lack of web-based cameras with better zoom quality, lack of a high-quality lighting source, a digital stethoscope, and lack of good quality microphones (Ward et al., 2012). Other than the infrastructure facilities, the platforms used for telepractice should be made secure in terms of client confidentiality and data storage (Doss et al., 2018; Emezue, 2020). These factors will address privacy concerns that pose risks to the code of ethics (Rogers, 2020).

Workplace/organisation-related Barriers in the use of Telepractice in the use of telepractice with

adults with dysphagia post-stroke

| The use of inexpensive web-based software with limited    |                          |
|---|--------------------------|
| privacy and security settings                             |                          |
| Lack of high-quality images for assessment                |                          |
| Lack of specially trained SLTs and staff for instrumental |                          |
| swallowing assessments like FEES and MBS for the          | Malandraki et al. (2011) |
| evaluation of people with dysphagia                       | _                        |
| Limited access to technology to conduct services online   |                          |
| Lack of assessment and treatment resources                |                          |
| Organisational policy restrictions                        | May and Erickson (2014)  |
| Privacy concerns posing risk to code of ethics            | Rogers (2020)            |
| Legal safeguards, safety precautions, criteria for client |                          |
| candidacy   | Miles et al. (2021)      |
| Technology specifications (in the implementation of       | Richmond et al. (2017)   |
| telepractice in the management of people with dysphagia)  |                          |
| Lack of detailed specifications or guidelines for         | Malandraki et al. (2021) |
| implementation of telepractice                            |                          |

In addition to the factors mentioned above, the lack of detailed specifications or

guidelines for the implementation of telepractice is also a significant workplace/organisation-

related barrier in the use of telepractice.

## 2.8 Beliefs of SLTs on the use of telepractice

This section reviews the general beliefs of SLTs on the use of telepractice. The opinions

of SLTs on the use of telepractice for the delivery of clinical services vary.

The different beliefs include:

(1) Telepractice is used satisfactorily in many forms of health care services. Similar view is suggested by Ward et al. (2014). Also, some SLTs consider telepractice as an effective service delivery platform compared to face-to-face therapy (Regina Molini-Avejonas et al., 2015; Sharma et al., 2013; Swales et al., 2020).

- (2) SLTs believed they developed a good rapport with their clients while providing clinical service for dysphagia via telepractice. Also, it was stated that the telepractice system was easy to use (Sharma et al., 2013). In addition, the clients also felt comfortable while conducting their swallow assessment via telepractice.
- (3) Some SLTs consider that telepractice could not be taken as a complete replacement for in-person clinical service delivery. The same was stated by Reynolds et al. (2009).
- (4) SLTs thought that the infrastructure needed for offering telepractice was entirely different from conventional service delivery, as reported in a survey study conducted in India (Mohan et al., 2017). However, the survey respondents also stated that telepractice could be made affordable, effective, and reliable with adequate technological infrastructure.
- (5) Majority of the SLTs felt telepractice was an appropriate clinical service delivery method for individuals with Parkinson's disease (Swales et al., 2020).
- (6) Telepractice can be considered an essential clinical service delivery platform in the care of adults with dysphagia. Evidence supports the use of telepractice as an equivalent alternative to in-person dysphagia care for adults in both inpatient and outpatient settings as reported by Malandraki et al. (2021).

#### **2.9** Conclusion

To conclude, this chapter reviewed the available literature concerning the use of telepractice in general and with adults with dysphagia post-stroke in particular. The review started with the definitions of telepractice and related terms, the types, technology involved, benefits and outcome, and the global perspectives in general and concerning telepractice in SLT and individuals with dysphagia. It also reviewed COVID-19 impacts on the use of telepractice.

The benefits, facilitators, and barriers of using telepractice with respect to the client, clinician and workplace were also detailed. Finally, it outlined the beliefs of SLTs on the use of telepractice.

The literature suggested that telepractice is a viable service delivery model across various client populations including adults with dysphagia. The importance of the use of telepractice under different clinical set up and contexts, including COVID-19, has been elaborated. There is lack of sufficient information on the use of telepractice especially with adults with dysphagia post-stroke in New Zealand. This warrants a study under New Zealand context on the perception of SLTs on the use of telepractice, specifically with adults with dysphagia post-stroke, as is presented in the next chapter. This is the first study to explore this in the New Zealand context.
#### **Chapter Three: Methodology**

#### **3.1 Introduction**

This chapter outlines the methodology used in the study starting with the research question and the chosen research design. Then, details are provided about the participants involved and the process of recruitment of participants. The descriptions of the methods used to collect, analyse and interpret the data are detailed. Finally, the ethical considerations of the study are explained.

#### 3.2 Research aims

The specific aims of the study include:

(1) To explore the perceptions of SLTs in New Zealand with respect to the use of telepractice with adults with dysphagia post-stroke

(2) To investigate the perceived benefits, facilitators, barriers and beliefs in the use of telepractice

#### 3.3 Research design

A cross-sectional, predominantly quantitative survey design was employed in this study. A cross-sectional survey design is a common design generally used to examine participants' attitudes, beliefs, and practices (Creswell & Guetterman, 2019). The design was specifically chosen to capture the perceptions of the participants on the use of telepractice.

#### **3.4 Participants**

Participants were SLTs working in New Zealand with experience using telepractice with adults with dysphagia post-stroke, recruited through the New Zealand Speech-Language Therapists' Association (NZSTA) and the Massey University Speech and Language Therapy Programme Facebook page (<u>https://www.facebook.com/newzealandspeechlanguagetherapy</u>). An email containing a brief introduction to the study (Appendix A), the information sheet (Appendix B) and the link to the survey were distributed to the organisation.

The inclusion criteria involved in the recruitment of participants of the study was the following:

The participant should be an SLT working in New Zealand who has experience on telepractice with adults with dysphagia post-stroke.

#### **3.5 Survey development**

The questions were drawn from a variety of sources including the content of previous surveys focused on the use of telepractice in the management of dysphagia and other neurological problems (Ayanikalath, 2017). A study on telepractice in people with Parkinson's disease (Swales et al., 2020) was particularly useful. This study specifically informed the development of a questionnaire for eliciting the perceptions of SLTs on the use of telepractice with people with Parkinson's disease. The survey questions were added on to the survey platform Qualtrics (discussed below).

The questionnaire contained a total of 10 items. These included:

 Participants' demographic information with seven questions: the number of years of experience of the SLTs, the highest level of qualification in SLT, level of education at which the participants learned about dysphagia, professional learning and development undertaken in telepractice by participants, the region in New Zealand where they are based, employment status and the facility where clinical services are offered to adults with dysphagia post-stroke.

- 2. Clinical services for adults with dysphagia post-stroke with two questions: the number of adults with dysphagia post-stroke treated by the participants during the last two years and the age group of adults with dysphagia post-stroke with whom the participants worked.
- Two questions about confidence with technology: the confidence level in using technology in general and in clinical practice.
- 4. Telepractice services for adults with dysphagia post-stroke with four questions: clinical services offered to adults with dysphagia post-stroke via telepractice; the platforms used, items used to augment clinical swallow evaluation via telepractice, and client-related factors considered when deciding to offer adults with dysphagia post-stroke clinical services via telepractice.
- 5. Expansion of clinical services for adults with dysphagia post-stroke via telepractice with two questions: whether they had a plan to expand clinical services for adults with dysphagia post-stroke via telepractice and the methods of developing clinical services.
- 6. Benefits of clinical services for adults with dysphagia post-stroke via telepractice: client-related, clinician-related and workplace/organisation-related benefits.
- 7. Facilitators of clinical services for adults with dysphagia post-stroke via telepractice: client-related, clinician-related and workplace/organisation-related facilitators.
- 8. Barriers of clinical services for adults with dysphagia post-stroke via Telepractice: clientrelated, clinician-related and workplace/organisation-related barriers.
- 9. Beliefs of participants about telepractice: using the level of agreement of the participants on 10 statements captured on a five-point scale (strongly disagree, somewhat disagree, neither agree nor disagree, somewhat agree, and strongly agree).
- 10. Additional comments with one question.

The format of the questions varied and comprised multiple-choice questions, binary yesno option questions, Likert scale type responses and open-ended responses. Likert scale type questions use fixed choice response format and are used to measure the attitudes or opinions of the participants (Burns & Grove, 2005). The open-ended questions avoid the bias that exists with pre-determined categories (Maxwell & Satake, 2006) and enable a detailed description of the participants' experience (Bryman, 2016). Strategies for constructing good questions were followed in formulating the survey questions (Creswell & Guetterman, 2019). These strategies included using positive wording, not using jargon, ensuring clarity of questions, asking one question at a time, and ensuring that the answers match the questions in multiple-choice questions.

#### **3.5.1 Pilot Study**

A pilot study was conducted by sending a copy of the questionnaire to an expert in telepractice for adults with dysphagia post-stroke in New Zealand with the intention of improving clarity in the questions. It is a recommended best practice to conduct a pilot study of a survey to determine if the sample population can understand the questions, complete the study, and check for any flaws (Ary et al., 2018; Punch, 2013). Pilot study will also help determine whether the survey questions adequately measure what was intended (Ary et al., 2018).

The expert was asked to give detailed feedback about the structure and comment on the questions' relevance and comprehensibility and this feedback was incorporated into the final survey questionnaire to improve clarity and, as a result, reframed six questions. Other changes involved addition of factors when selecting adults with dysphagia post-stroke for clinical services via telepractice (technology skills, access to hardware, access to software, internet quality and access to support person). Finally, questions regarding benefits, facilitators and

barriers of clinical services were altered; here, sub-titles such as client-related factors, clinician related factors and workplace/organisation-related factors were added.

#### **3.6 Data Collection**

An anonymous online survey (see Appendix C) was used to collect information from the participants over three weeks via Qualtrics. After two weeks, a follow-up reminder was sent to the organisation, asking them to remind SLTs who met the inclusion criteria to complete the survey.

An online survey is cost-efficient, easy to distribute and access, and allows the participants to complete the survey at their own pace in a place comfortable to them (Dillman et al., 2014). It also considerably reduces the data input time (Irwin et al., 2019). Another advantage of online surveys is guarantee of complete anonymity. Another contextual reason that could be attributed to the use of online surveys is lockdown due to the COVID-19 pandemic, which meant participants could not have been interviewed in person.

Qualtrics was used to design the questionnaire as it is one of the survey platforms that enable researchers to construct and administer surveys quickly. Qualtrics is a web-based survey tool to conduct survey research and data collection in which survey can be prepared, sent and analysed as suggested by California State University Long Beach (Albaum & Smith, 2006; CSULB, 2020).

#### **3.7 Data Analysis**

The study used basic descriptive statistics to analyse the data including mean and used frequency counts, frequency tables and graphs to depict the data. Open-ended questions were analysed using a data-driven coding approach (Gibbs, 2015). The processes involved were as follows: (1) reading the responses provided by the participants (2) categorising similar responses

into codes (3) organising codes into themes (4) presenting the number of participants who provided a response under each category in a summary table along with the theme.

#### **3.8 Ethical Considerations**

The research project was evaluated by peer review and judged low risk, and a low-risk notification to the Massey University Human Ethics Committee was submitted (Ethics notification number: 4000024328; (see Appendix D). The study aims and procedures were outlined in an information sheet.

Two of the key ethical issues considered were consent and confidentiality. Due to the confidential nature of the study, once the participants chose to complete the survey, consent to participate in the study was implied, which was explained in the information sheet. The information sheet also stated no anticipated risks involved in participating in the survey.

This survey will maintain the privacy and confidentiality of the participants as the survey collected no identifying information such as the participants' name, age, gender, and address to maintain anonymity. Also, the researcher has not directly contacted any of the participants. Only the researcher and her supervisors will have access to the data. Electronic data is stored on a password-protected device and information will be kept for five years following the publication of the results. When disposed of, the University's confidential waste service for printed materials will be utilised.

#### **3.9 Conclusion**

This chapter outlined the aims of the research, design, participant recruitment, and data collection and analysis procedures. In addition, it described ethical considerations. The methodological considerations outlined justify the tools and techniques used for the study. A descriptive cross-sectional survey design was the best fit for the study because it permitted

analysis of the perceptions of the participants in the use of telepractice with adults with dysphagia post-stroke. The research design allowed for the actual perceptions and beliefs of the participants to be expressed based on the questionnaire.

#### **Overview of the methods**

First, literature on the use of telepractice in general and in particular to the management of dysphagia was collected. The research question, the aims of the study and ethical considerations were then formulated. The research design was then arrived up on. It was decided to conduct an online survey by means of Qualtrics tool to understand the perceptions of the SLTs who work with adults with dysphagia post-stroke with respect to the clients, clinicians and the workplace/organisation. The beliefs of SLTs on the use of telepractice were to be elicited by means of Likert scale questions. The questionnaire was developed with 10 items and a pilot study was conducted. The questionnaire was finalised and then sent to participants who are SLTs in New Zealand with experience in telepractice with adults with dysphagia post-stroke. The data obtained from six participants was subjected to analysis and meaningful outcomes were discussed.

## **Chapter Four: Results**

## **4.1 Introduction**

The purpose of this study was to explore the perceptions of speech-language therapists (SLTs) in New Zealand on the use of telepractice with adults with dysphagia post-stroke, along with the perceived benefits, facilitators, barriers, and beliefs using an online survey. This chapter presents the results of the study.

## 4.2 Survey results

Six SLTs in New Zealand with experience in telepractice with adults with dysphagia post-stroke participated.

## 4.2.1 Demographics of the participants

The demographics of the participants are explained as follows:

- SLTs' years of experience
- Highest level of qualification in speech and language therapy
- Level of education at which participant learned about dysphagia
- Professional learning and development undertaken in telepractice
- The region in New Zealand where participants are based
- Employment status and
- The facility where clinical services are provided to adults with dysphagia post-stroke.

## 4.2.1.1 SLTs' years of experience

The number of years the participants had practised as therapists varied. The responses ranged from three years to 26 years. Most of the participants (n=4) had between three to

seven years of experience, but two participants (1 and 6) had 26 and 25 years of experience respectively. The mean number of years of experience of the participants was 11.58 years. Figure 1 represents the number of years of experience of the participants.

## Figure 1





## 4.2.1.2 Qualification of the participants

The highest educational qualification of the participants in the study varied. Three participants had completed their undergraduate degree, followed by master's degree (n=2) and postgraduate certificate (n=1).

## 4.2.1.3 Learning about dysphagia

The level of education at which the participants learned about dysphagia varied. Three participants learned dysphagia at the undergraduate level of education, master's level (n=2), and postgraduate certificate level (n=1).

#### 4.2.1.4 Professional learning and development of participants in telepractice

The participants identified several professional learning and development opportunities they had engaged in relation to telepractice. All the participants (n=6) had undertaken selfdirected learning (reading books, journal articles) and online learning. The participants' other professional learning and development opportunities included engaging in peer learning (n=5) and attending conference, seminar, and workshops (n=5). One participant attended New Zealand telehealth webinars. None of the participants had training in telepractice during their undergraduate and postgraduate coursework. Figure 2 depicts the events of professional learning and development of the participants in telepractice.

#### Figure 2



Events of Professional Learning and Development of the Participants in Telepractice

The events of professional learning and development undertaken in telepractice by the participants varied, as seen above.

## 4.2.1.5 Regions where participants are based in New Zealand

The regions in New Zealand where the participants were based included Auckland (n=3), Whanganui-Manawatu (n=1), Hawke's Bay (n=1), and Gisborne (n=1). Of the 16 regions in New Zealand, the participants were from only four areas; the others are not represented.

## **4.2.1.6 Employment status of participants**

The employment status of the participants in the study included part-time employment (n=2) and full-time employment (n=4).

## 4.2.1.7 Facilities where clinical services are provided by participants to adults with

## dysphagia post-stroke

The facilities where clinical services are provided to adults with dysphagia post-stroke are presented in Figure 3.

## Figure 3



Facilities where Clinical Services Provided to Adults with Dysphagia post-stroke

## 4.2.2 Clinical services to adults with dysphagia post-stroke

The number and age group of adults with dysphagia post-stroke that the SLTs provided clinical services to in the last two years are described here.

## 4.2.2.1 Number of adults with dysphagia post-stroke seen by participants in the last two

years

Table 11 depicts the number of adults with dysphagia post-stroke seen by the participants during

the last two years.

## Table 11

Number of adults with dysphagia post-stroke seen by participants in the last two years

| Number of adults with dysphagia post- | Number of SLTs involved |
|---------------------------------------|-------------------------|
| stroke seen                           |                         |
| 30                                    | 1                       |
| 50                                    | 1                       |
| 75                                    | 1                       |
| More than 100                         | 3                       |

From the table, it is evident that the number of adults with dysphagia post-stroke treated by the participants during the last two years varied and included 30 (n=1), 50 (n=1), 75 (n=1), and 100 and above (n=3).

## 4.2.2.2 Age group of adults with dysphagia post-stroke seen by SLTs

Figure 4 includes the age group of adults with dysphagia post-stroke with whom the participants worked. The participants worked with adults across the age span. All the participants (n=6) worked with adults aged more than 40 years, whereas only five participants stated that they worked with clients in the age group of less than 40

## Figure 4



The Age Group of Adults with Dysphagia post-stroke seen by SLTs

## 4.2.3 Confidence with technology

This describes the level of confidence with technology in using technology in general and in clinical practice.

## 4.2.3.1 Level of confidence of participants in using technology in general

The participants' level of confidence in using technology, in general, was rated as

excellent (n=3) and good (n=3).

## 4.2.3.2 Level of confidence of participants in using technology in clinical practice

The level of confidence of participants in using technology in clinical practice was rated

as good (4) and excellent (2).

## 4.2.4 Telepractice service to adults with dysphagia post-stroke

This section describes clinical services offered to adults with dysphagia post-stroke via telepractice and the platforms used.

#### 4.2.4.1 Clinical services offered to adults with dysphagia post-stroke via telepractice

The clinical services offered to adults with dysphagia post-stroke varied and included assessment (n=5), education for clients and caregivers (n=5), direct intervention (exercises) (n=4), feeding trials (n=3), rehabilitation and compensatory strategies (muscle strength training) (n=4).

# 4.2.4.2 Platforms used for providing clinical services to adults with dysphagia post-stroke by the participants via telepractice

The platforms used by the participants when providing clinical services to adults with dysphagia post-stroke included audio calls such as telephone and Skype calls (n=3) and videoconferencing (n=5).

#### 4.2.4.3 Items used to augment clinical swallow evaluation via telepractice

No participants stated that they used any of the following items: cervical auscultation, pulse oximetry, surface electromyography, or others to augment clinical swallow evaluation via telepractice.

# 4.2.4.4 Client-related factors considered when deciding to offer adults with dysphagia poststroke clinical services via telepractice

The participants considered several client-related factors when deciding to offer telepractice to adults with dysphagia post-stroke including: clinical cognitive behavioural ability such as cognitive functioning, multitasking, attention (n=6); physical/sensory ability such as hearing, vision, manual dexterity, sitting tolerance (n=5); communication ability such as auditory comprehension, literacy, speech intelligibility (n=6); technology skills (n=6); access to hardware such as laptop, iPad (n=6); access to software such as skype, zoom (n=4); internet quality (n=5); access to support person such as caregiver (n=5).

## 4.2.5 Expansion of clinical services for adults with dysphagia post-stroke via telepractice

This section describes the plan to expand clinical services to adults with dysphagia poststroke via telepractice by the participants and the expansion methods.

## 4.2.5.1 Plan and methods of expanding clinical services to adults with dysphagia post-

#### stroke

The majority of the participants (n=4) planned to expand their clinical services to adults with dysphagia post-stroke by increasing the number of clients. The rest of the participants (n=2) planned to expand their services by providing communication therapy to clients. However, none of the participants reported growing by increasing the range of equipment or services.

## 4.2.6 Benefits of clinical services for adults with dysphagia post-stroke via telepractice

The benefits of clinical services for adults with dysphagia post-stroke via telepractice in relation to client, clinician and workplace/organisation are described in this section. These benefits were coded and then organised into themes. The themes are presented in Table 12 and described in more detail in the following sections.

#### Table 12

| Themes                            | Codes included                          |
|-----------------------------------|---|
| Client-related benefits           |   |
| COVID-19 and other health factors | Lockdown situations                     |
|                                   | Safety proofing for future lockdowns    |
|                                   | Less exposure to hospital viruses       |
| Therapy access and logistics      | Therapy access                          |
|                                   | Reduced travel                          |
|                                   | Reduced travel costs                    |
|                                   | No parking                              |
|                                   | Reduced cost for private therapy        |
| Clinical service delivery         | Seen quicker                            |
|                                   | Session frequency and/or duration       |
|                                   | Therapy duration                        |
| Clinician-related benefits        |   |
| Clinical service delivery         | Talk with caregivers; See more patients |

Themes identified under the benefits of telepractice

|   | More clinical time                   |
|---|--------------------------------------|
|   | Quality of work                      |
|   | Reduced caseloads                    |
|   | Reduced waitlists                    |
| Work setting                            | Work from home, Sessions from office |
|   | No need to book clinic room          |
|   | Use of trained therapy assistant     |
| COVID-19 factors                        | Lockdown situations                  |
| Reduced travel and costs                | Reduced travel time and requirements |
|   | Reduced cost                         |
| Workplace/organisation-related benefits |                                      |
| Clinical service delivery               | Reduced waitlists                    |
|   | Equitable access                     |
|   | Quality service                      |
|   | Ministry of Health targets           |
| Reduced travel and costs                | Travel time ; Reduced cost           |
|   | Environmental factors                |
| Employee quality of life                | Employee quality of life             |

## 4.2.6.1 Client-related benefits

The participants listed the client-related benefits of clinical services for adults with dysphagia post-stroke via telepractice. The participants' responses were coded as illustrated in Table13 and organised into three themes: COVID-19 and other health factors, therapy access and logistics, and clinical service delivery.

## Table 13

## Code Summary for Client-related Benefits

| Theme                | Code                | Number  | Example                                |
|----------------------|---------------------|---------|--|
|                      |                     | of SLTs |  |
|                      | Lockdown situations | 1       | "It is really helpful in lockdown      |
|                      |                     |         | situations"                            |
| COVID-19 and         | Safety proofing for | 1       | "Safety proofing for on-going support  |
| other health factors | future lockdowns    |         | and assessments in the event of future |
|                      |                     |         | lockdowns"                             |
|                      | Less exposure to    | 1       | " less exposure to hospital viruses"   |
|                      | hospital viruses    |         |  |
|                      | Therapy access      | 2       | "More accessible therapy for patients  |
|                      |                     |         | who may not have been able to, or want |

|                  |                         |   | to, come into a clinic or someone visit |
|------------------|-------------------------|---|---|
| Therapy access   | Reduced travel          | 2 | "No need to travel"                     |
| and logistics    | Reduced travel costs    | 1 | " reduces travel costs"                 |
|                  | No parking              | 1 | " not having to find parking"           |
|                  | Reduced cost for        | 1 | " reduced costs for private practice    |
|                  | private therapy         |   | consults"                               |
|                  | Seen quicker            | 1 | "They may be seen quicker as SLTs       |
| Clinical service |                         |   | time may be freed up"                   |
| delivery         | Session frequency and / | 3 | "More frequent sessions such as 3 x 15  |
|                  | or duration             |   | minutes sessions each week"             |
|                  | Therapy duration        | 1 | "Intervention can be completed in a     |
|                  |                         |   | more timely manner"                     |

The theme COVID-19 and other health factors included three codes: lockdown situations, safety proofing for future lockdowns and less exposure to hospital viruses. One each of the participants suggested that telepractice is helpful in lockdown situations and acts as safety proofing for on-going support and assessment in future lockdowns. One participant also suggested that one of the client-related benefits of using telepractice could be less exposure to hospital viruses.

The theme therapy access and logistics included five codes: therapy access, reduced travel, reduced travel costs, no parking, and reduced cost for private practice. Two participants suggested therapy access and reduced travel. One of the participants indicated the rest of the codes, such as reduced travel costs, no parking, and lowered costs for private practice. As far as therapy access was concerned, the participants thought that telepractice helped to have more accessible therapy for individuals who may not have been able to, or want to, come into a clinic or someone visit their home.

The last theme service delivery comprised three codes as seen quicker, session frequency and/or duration and therapy duration. The response for seen quicker was that it is possible to see the clients quicker as the SLTs may get their time freed up with telepractice. It is also anticipated

that interventions could be completed in a timely manner and more frequent sessions could be conducted.

#### 4.2.6.2 Clinician-related benefits

The participants were asked to list the clinician-related benefits of clinical services for adults with dysphagia post-stroke via telepractice. The participants' responses were coded as illustrated in Table 14 and organised into four themes: clinical service delivery, work setting, COVID-19 factors; and reduced travel and costs.

The theme clinical service delivery was organised into six codes as talk with caregivers; treat more patients; more clinical time; quality of work; caseloads; and waitlists. Out of these, the statement, see more patients, was suggested by three participants because of the ability to see more patients within a day. Two participants suggested more clinical time as clinician-related benefits of using telepractice. The other benefits of clinician-based telepractice under the theme clinical service delivery included improved quality of work due to time and energy conserved; reduced caseloads and reduced waitlists.

The next theme, work setting, was organised into four codes: work from home, sessions from office, no need to book clinic room, and use of trained therapy assistant at patient's home. These were suggested by one participant each respectively.

## Table 14

| Theme                        | Code                             | Number<br>of SLTs | Example   |
|------------------------------|----------------------------------|-------------------|---|
| Clinical service<br>delivery | Talk with caregivers             | 1                 | "Ability to talk with caregivers at<br>private hospitals"                 |
|                              | See more patients                | 3                 | "Ability to see more patients within a day"                               |
|                              | More clinical time               | 2                 | "Increasing clinical time"  |
|                              | Quality of work                  | 1                 | "improved quality of work due to time<br>and energy conserved"            |
|                              | Reduced caseloads                | 1                 | " keeps caseloads down"   |
|                              | Reduced waitlists                | 1                 | " keeps waitlists down"   |
| Work setting                 | Work from home                   | 1                 | "Ability to work from home"   |
|                              | Sessions from office             | 1                 | "Can do sessions from office"   |
|                              | No need to book clinic room      | 1                 | "Don't have to book clinic room"  |
|                              | Use of trained therapy assistant | 1                 | " we can sometimes use a trained<br>therapy assistant at patient's home:" |
| COVID-19 factors             | Lockdown situations              | 1                 | "it is really helpful in lockdown situations"                             |
| Reduced travel and costs     | Reduced travel time requirements | 2                 | " there is reduced travel time within the community"                      |
|                              | Reduced cost                     | 1                 | "reduced cost"  |

## Code Summary for Clinician-related Benefits

The last theme included reduced travel and costs which comprised two codes such as travel time requirements and reduced cost. Of these reduced travel time requirements was suggested by two participants.

## 4.2.6.3 Workplace/organisation-related benefits

The participants listed the workplace/organisation-related benefits of clinical services for adults with dysphagia post-stroke via telepractice. The participants' responses were coded as illustrated in Table 15 and organised into three themes: clinical service delivery, reduced travel and costs, and employee quality of life.

## Table 15

| Theme                     | Code                  | Number  | Example                                    |
|---------------------------|-----------------------|---------|--|
|                           |                       | of SLTs |  |
| Clinical service delivery | Reduced waitlists     | 2       | "potential to keep waitlists down"         |
|                           | Equitable access      | 1       | "More equitable access for more patient    |
|                           |                       |         | (if barriers to assessing such as internet |
|                           |                       |         | and devices are removed"                   |
|                           | Quality service       | 1       | "high quality, clear and simple            |
|                           |                       |         | service as there is not much room for      |
|                           |                       |         | 'small talk' or hiding really!"            |
|                           | Ministry of Health    | 1       | "meet Ministry of Health targets"          |
|                           | targets               |         |  |
| Reduced travel and        | Reduced travel time   | 1       | "Less travel time"                         |
| costs                     | Reduced cost          | 2       | "Less cost of petrol"                      |
|                           | Environmental factors | 1       | "Reduced environmental factors             |
|                           |                       |         | regarding travel"                          |
| Employee quality          | Employee quality of   | 1       | "Healthier, happier employees, better      |
| of life                   | life                  |         | work-life balance and general quality of   |
|                           |                       |         | life"                                      |

Code Summary for Workplace/organisation-related Benefits

The theme of clinical service delivery comprised four codes as reduced waitlists, equitable access; quality service; and Ministry of Health targets. Out of these, the statement, reduced waitlists, was suggested by two participants. More equitable access for more patients (if barriers to assessing such as internet and devices are removed) was mentioned by one participant. Telepractice results in high quality, clear and simple service as there is not much room for 'small talk' or hiding, was the statement of another participant. The other information was concerning meeting Ministry of Health targets.

The second theme was reduced travel and costs, which was organised into three codes: reduced travel time, reduced cost, and reduced environmental factors regarding travel. Of these, reduced cost was mentioned by two participants and the others by one participant each. The reason for reduced cost included less cost of petrol as there is less travel.

The last theme was employee quality of life, which one participant put forward mainly because of the perception that telepractice resulted in healthier and happier employees, better work-life balance and general quality of life.

## 4.2.7 Facilitators of clinical services for adults with dysphagia post-stroke via

#### telepractice

The participants were asked to list the client, clinician and workplace/organisation-related facilitators of clinical services for adults with dysphagia post-stroke via telepractice. These facilitators were coded and then organised into themes. The themes identified under the facilitators of telepractice are presented Table 16 and described in more detail in the following sections.

#### Table 16

| Themes                             | Codes included              |
|------------------------------------|-----------------------------|
| <b>Client-related facilitators</b> |                             |
| Devices                            | Telephone support           |
|                                    | iPad                        |
|                                    | Video call device           |
| Technology                         | Technology                  |
|                                    | Familiarity with technology |
| Support person                     | Support person              |
|                                    | Supportive partner          |

Themes Identified under the Facilitators of Telepractice

| Willingness                                 | Willingness                          |
|---|--------------------------------------|
| Clinician-related facilitators              |                                      |
| Training                                    | Training and Experience              |
|   | Positive feedback                    |
| Finding solutions                           | Problem solving                      |
|   | Talking as a team                    |
|   | Flexibility and open minded approach |
| Experience with technology                  | Comes in handy                       |
|   | To connect                           |
|   | Confidence                           |
| Workplace/organisation-related facilitators |                                      |
| Work setting                                | Telephone booths                     |
|   | Work from home                       |
|   | Quiet room                           |
| Support by organisation                     | Organisation support                 |
|   | Posters                              |

## 4.2.7.1 Client-related facilitators

The participants were asked to list the client-related facilitators of clinical services for

adults with dysphagia post-stroke via telepractice. The participants' responses were coded as

illustrated in Table 17 and organised into four themes: devices, technology; support person; and

willingness.

## Table 17

Code Summary for Client-related Facilitators

| Themes         | Code                  | Number  | Example   |
|----------------|-----------------------|---------|---|
|                |                       | of SLTs |   |
| Devices        | Telephone support     | 1       | "Telephone support to set up initially"                         |
|                | iPads                 | 1       | " loan iPads"   |
|                | Internet access       | 2       | "Access to internet"  |
|                | Video call experience | 1       | "Video call experience with family"                             |
| Technology     | Technology skills     | 1       | "technology skills"   |
|                | Familiarity           | 1       | "Familiarity with technology"                                   |
| Support person | Support person        | 1       | "Support person to facilitate"                                  |
|                | Supportive partner    | 1       | "Supportive partner husband willing to help<br>get them set up" |
| Willingness    | Willingness           | 1       | "Willingness to give it a try"                                  |

Two participants stated that the most frequently identified client-related facilitator regarding the theme devices was internet access. The responses included access to internet and gadgets. In addition, one of the participants suggested telephone support, iPads on loan, and video call experience with family members as other client-related facilitators in using telepractice under the theme devices.

The second theme technology comprised two codes: technology skills and familiarity. Technology skills and understanding of technology were considered as client-related facilitators and were proposed by one each of the participants.

The third theme identified was support person and was organised into two codes: support person and supporting partner. Support person to facilitate and a supportive husband willing to help set up were the statements suggested by one each of the participants.

The last theme was willingness, which included giving telepractice a try. One participant stated this.

#### 4.2.7.2 Clinician-related facilitators

The participants were asked to list the clinician-related facilitators of clinical services for adults with dysphagia post-stroke via telepractice. The participants' responses were coded as illustrated in Table 18 and organised into three themes: training, finding solutions, and experience with technology.

The theme training was organised into three codes: training, experience, and positive feedback. Two participants suggested training to use zoom. One example was providing positive feedback after a session was considered a clinician-related facilitator by one participant. Having

experience was identified as a clinician-related facilitator as "it comes in handy when trying to explain to clients how to connect if they are unfamiliar".

The second theme was finding solutions; this was organised into four codes as problemsolving, team discussion, flexibility, and an open-minded approach. Among these, problemsolving was mentioned by two participants. Examples included a willingness to be available for problem-solving over the phone. Talking as a team regarding success, flexibility, and openminded approach were the other clinician-related facilitators in using telepractice as suggested by one each of the participants.

## Table 18

| Themes                     | Code                 | Number  | Example   |
|----------------------------|----------------------|---------|---|
|                            |                      | of SLTs |   |
|                            | Training             | 2       | " training how to use zoom"   |
| Training                   |                      |         |   |
|                            | Experience           | 1       | "Experience with zoom and training how to<br>use it"                                      |
|                            | Positive feedback    | 1       | "providing positive feedback  |
|                            |                      |         | after a session like "thanks for the session, it<br>went really well, I got all I needed" |
|                            | Problem solving      | 2       | "Willingness to being available for problem   |
| Finding                    |                      |         | solving over the phone"   |
| solutions                  | Talking as a team    | 1       | "talking as a team regarding success"   |
|                            | Flexibility          | 1       | "Flexibility"   |
|                            | Open minded approach | 1       | "Open minded approach"  |
| Experience with technology | Comes in handy       | 1       | "This comes in handy when trying to explain to a patient"                                 |
|                            | To connect           | 1       | " how to connect if they are unfamiliar"  |
|                            | Confidence           | 2       | "Use of a buddy system to improve confidence"   |

Code Summary for Clinician-related Facilitators

The last theme was experience with technology and was organised into three codes as it comes in handy, to connect and confidence. As far as confidence in using technology was concerned, the primary response included utilising a buddy system to improve morale. Statements suggesting clinician-related facilitators under the theme experience with technology also included being able to support the person to connect if he is unfamiliar.

#### 4.2.7.3 Workplace/organisation-related facilitators

The workplace/organisation-related facilitators of clinical services for adults with dysphagia post-stroke via telepractice were collected and analysed. The participants' responses were coded as illustrated in Table 19 and organised into two themes: work setting and support by organisation.

The participants' most frequently identified workplace/organisation-related facilitators under the theme work setting include worked from home; suggested by two participants each. The other workplace/organisation-related facilitators included telephone booths and a quiet room dedicated to telepractice in the workplace.

## Table 19

| Theme        | Code             | Number  | Example                                 |
|--------------|------------------|---------|---|
|              |                  | of SLTs |   |
| Work setting | Telephone booths | 1       | "Telephone booths are great"            |
|              | Work from home   | 2       | "Allowance to work from home one day a  |
|              |                  |         | week"                                   |
|              | Quiet room       | 1       | " quiet room dedicated to TH in the     |
|              |                  |         | workplace"                              |
| Support by   | Organisation     | 1       | " supported by the organisation".       |
| organisation | support          |         |   |
|              | Posters          | 1       | "posters encouraging use of telehealth" |

Code Summary for Workplace/Organisation-related Facilitators

The other theme of organisation-related facilitator comprised two codes: organisation support and posters. These were suggested by one participant each. For example, the reason included telepractice was endorsed by the organisation, and posters encouraged the use of telepractice.

## 4.2.8 Barriers of clinical services for adults with dysphagia post-stroke via telepractice

The client, clinician and workplace/organisation-related barriers of clinical services for adults with dysphagia post-stroke via telepractice were coded and then organised into themes. The themes are presented in Table 20 and described in more detail in the following sections.

## Table 20

| Themes                                  | Codes included                   |  |  |  |
|---|----------------------------------|--|--|--|
| Client-related barriers                 |                                  |  |  |  |
| Technology                              | Technology set up;               |  |  |  |
|   | Technology skills                |  |  |  |
|   | Technology familiarity           |  |  |  |
|   | Internet                         |  |  |  |
| Clinical service                        | Preference for in-patient visits |  |  |  |
|   | Reduced rapport                  |  |  |  |
|   | Lack of support person           |  |  |  |
|   | Reluctance                       |  |  |  |
| Patient character                       | Cognitive impairment             |  |  |  |
| Clinician-related barriers              |                                  |  |  |  |
| Technology                              | Fear of technology               |  |  |  |
|   | Familiarity                      |  |  |  |
| Mind set                                | Dislike of change                |  |  |  |
|   | Preference for in-person meeting |  |  |  |
|   | Reluctance                       |  |  |  |
|   | Prefer face-to-face              |  |  |  |
|   | Willingness                      |  |  |  |
| Therapy access                          | Quality of intervention          |  |  |  |
|   | Ability to talk to client        |  |  |  |
|   | Effort and/or skill              |  |  |  |
| Workplace/organisation-related barriers |                                  |  |  |  |
| Therapy                                 | Objective assessment             |  |  |  |

Themes Identified under the Barriers of Telepractice

| Work setting | Quiet clinic room<br>Lack of space             |
|--------------|--|
| Devices      | Lack of shared laptop<br>Poor internet quality |

## 4.2.8.1 Client-related barriers

The participants listed the client-related barriers of clinical services for adults with dysphagia post-stroke via telepractice. Under the theme technology, were four codes; these included technology set up, technology skills, technology familiarity and the internet. The most frequently identified client-related barrier (n=4) was poor internet quality. Poor technology skill was another barrier identified by two participants. Poor technology skill was another barrier suggested by two participants.

The responses were coded as illustrated in Table 21 and organised into three themes:

technology, clinical service, and patient character.

## Table 21

| Themes           | Code               | Number<br>of SLTs | Example   |
|------------------|--------------------|-------------------|---|
| Technology       | Technology set up  | 1                 | "Use of technology set up"  |
|                  | Technology skills  | 2                 | "Poor technology skills"  |
|                  | Technology         | 1                 | "Lack of familiarity with technology"                                     |
|                  | familiarity        |                   |   |
|                  | Internet quality   | 4                 | " poor internet quality"  |
| Clinical service | Preference for in- | 1                 | 'preference for in-person visits when                                     |
|                  | patient visits     |                   | COVID-19 has not been an issue"   |
|                  | Reduced rapport    | 1                 | "reported reduced rapport at times"                                       |
|                  | Lack of support    | 1                 | "Lack of support person to facilitate"                                    |
|                  | person             |                   |   |
|                  | Reluctance         | 1                 | " being reluctant to giving it a go - saying<br>"I'm a bit of a dinosaur" |

Code Summary for Client-related Barriers

| Patient character | Cognitive impairment | 1 | "cognitive impairment" |
|-------------------|----------------------|---|------------------------|
|                   |                      |   |                        |

Poor technology skill was stated as one of the barriers by two participants under the theme of technology. The other barriers under the theme of technology included lack of familiarity with technology and the use of technology set up and reported by one participant each. Reluctance to give telepractice a go was suggested as one of the barriers in using telepractice by one participant.

## 4.2.8.2 Clinician-related barriers

The clinician-related barriers of clinical services for adults with dysphagia post-stroke via telepractice were organised into three themes, as detailed in Table 22.

## Table 22

| Theme          | Code                | Number<br>of SLTs | Example                               |
|----------------|---------------------|-------------------|---------------------------------------|
|                | Fear of technology  | 1                 | "Fear of technology"                  |
| Technology     | Lack of familiarity | 1                 | "Lack of familiarity with             |
|                |                     |                   | technology"                           |
|                | Dislike of change   | 1                 | "Dislike of change"                   |
| Mind set       | Preference for in-  | 1                 | "Preference for in-person meeting /   |
|                | person meeting      |                   | connection"                           |
|                | Reluctance          | 1                 | "Reluctance to give it a go"          |
|                | Prefer face-to-face | 1                 | "prefer face-to-face"                 |
|                | Lack of Willingness | 1                 | "Lack of willingness to change        |
|                |                     |                   | normal practice"                      |
|                | Fear of quality of  | 1                 | "fear of quality of intervention      |
| Therapy access | intervention        |                   | being compromised"                    |
|                |                     |                   |                                       |
|                | Ability to talk to  | 1                 | "Ability to talk to client through IT |
|                | clients             |                   | issues without being there            |
|                |                     |                   | physically"                           |
|                | Effort and/or skill | 1                 | "Effort and/or skill required to      |
|                |                     |                   | support clients with technology       |
|                |                     |                   | issues"                               |

## Code Summary for Clinician-related Barriers

Under the theme technology, two codes included fear of technology and lack of familiarity. The second theme was mind set and included: dislike of change, preference for an inperson meeting, reluctance, prefer face-to-face, and willingness. Under the third theme therapy access, three codes included fear of the quality of intervention, ability to talk to the clients, and effort and/or skill required to support clients with technology issues.

#### 4.2.8.3 Workplace/organisation-related barriers

The workplace/organisation-related barriers of clinical services for adults with dysphagia post-stroke via telepractice were organised into three themes, as detailed in Table 23.

#### Table 23

| Theme        | Code                    | Number of<br>SLTs | Example  |
|--------------|-------------------------|-------------------|--|
| Therapy      | Objective<br>assessment | 1                 | "We can't do objective measures via<br>telepractice such as cough reflex<br>testing" |
| Work setting | Quiet clinic room       | 1                 | "Access to quiet clinic room"  |
|              | Lack of space           | 1                 | "Lack of space to work in a quiet room"  |
| Devices      | Lack of shared laptop   | 1                 | "Access to shared laptop"  |
|              | Poor internet quality   | 1                 | " lack of good quality internet"   |

Code Summary for Workplace/Organisation-related Barriers

Under the theme therapy, was included one code objective assessment referring to an example of cough reflex testing. The second theme worked setting included two codes: quiet clinic room and lack of space. The third theme under workplace/organisation-related barriers was devices and was organised into two codes: lack of shared laptop and poor internet quality.

## **4.3.** Beliefs about telepractice

The participants indicated on a five-point Likert scale whether they agreed or disagreed

with a series of 10 statements about telepractice. The participants' responses are presented in

Table 24.

## Table 24

## Frequency counts of the beliefs of SLTs about the use of telepractice

| Sl. | Statement                                      | Responses |     |     |     |    |
|-----|--|-----------|-----|-----|-----|----|
| No. |  |           | 1   |     |     | 1  |
|     |  | SD        | SwD | NDA | SwA | SA |
| 1   | Telepractice is an appropriate method to       | 0         | 0   | 1   | 3   | 2  |
|     | deliver clinical services for adults with      |           |     |     |     |    |
|     | dysphagia post-stroke                          |           |     |     |     |    |
| 2   | Telepractice will eventually become a          | 0         | 3   | 1   | 0   | 2  |
|     | mainstream delivery method for speech-         |           |     |     |     |    |
|     | language therapy services for adults with      |           |     |     |     |    |
|     | dysphagia post-stroke                          |           |     |     |     |    |
| 3   | In 10 years, most adults in New Zealand        | 0         | 0   | 2   | 1   | 3  |
|     | with dysphagia post-stroke will have access    |           |     |     |     |    |
|     | to clinical services via telepractice          |           |     |     |     |    |
| 4   | Over the next 10 years, older client's limited | 1         | 1   | 2   | 1   | 1  |
|     | familiarity and interest in technology will    |           |     |     |     |    |
|     | become less of a barrier to telepractice       |           |     |     |     |    |
| 5   | I will only use telepractice if in-patient     | 1         | 3   | 0   | 1   | 1  |
|     | delivery is not an option                      |           |     |     |     |    |
| 6   | You cannot develop the same level of client-   | 0         | 5   | 0   | 0   | 1  |
|     | clinician rapport via telepractice as is       |           |     |     |     |    |
|     | achieved in-person                             |           |     |     |     |    |
| 7   | My workplace/organisation is open and          | 0         | 0   | 2   | 2   | 2  |
|     | interested in using telepractice to provide    |           |     |     |     |    |
|     | services for adults with dysphagia post-       |           |     |     |     |    |
|     | stroke   |           |     |     |     |    |
| 8   | Telepractice can be used to provide a service  | 0         | 1   | 2   | 1   | 2  |
|     | that is as effective as in-person delivery     |           |     |     |     |    |
| 9   | Telepractice will only benefit adults with     | 3         | 2   | 1   | 0   | 0  |
| -   | dysphagia post-stroke who live in regional     | _         |     |     |     | -  |
|     | towns or rural areas, not city centres         |           |     |     |     |    |
| 10  | Telepractice has the potential to improve      | 0         | 1   | 1   | 3   | 1  |
|     | the level of care I am currently providing for | -         | -   | -   | -   | -  |
|     | adults with dysphagia post-stroke              |           |     |     |     |    |

(Key SD: Strongly disagree, SwD: Somewhat disagree, NDA: neither disagree nor agree;

#### SwA: Somewhat agree; SA: Strongly agree)

#### 4.3.1 Levels of agreement of SLTs

The first statement, telepractice is an appropriate method to deliver clinical services to adults with dysphagia post-stroke was agreed upon by five out of six participants. The second statement that telepractice will eventually become a mainstream delivery method for speechlanguage therapy services for adults with dysphagia post-stroke had a mixed response with half of the participants somewhat disagreeing, one participant neither disagreeing nor agreeing, and two participants strongly agreeing.

For the third statement that in 10 years, most adults in New Zealand with dysphagia poststroke will have access to clinical services via telepractice, three participants strongly agreed and one participant somewhat agreed. Two participants neither agreed nor disagreed.

The fourth statement that over the next 10 years, older clients' limited familiarity and interest in technology will become less of a barrier in the use of telepractice had a mixed response, with two participants agreeing, two participants disagreeing and two neither agreeing nor disagreeing.

The fifth statement that telepractice will be used only if inpatient delivery is not an option was agreed upon only by two participants, whereas four participants disagreed with this statement.

The majority of the participants (five out of six) somewhat disagreed with the sixth statement that they could not develop the same level of client-clinician rapport via telepractice as is achieved in person. Only one participant strongly agreed with this statement.

The seventh statement that the participants' workplace/organisation was open and interested in using telepractice to provide services for adults with dysphagia post-stroke was agreed on by four out of six participants, whereas, two participants neither agreed nor disagreed.

The response to the eighth statement was that one-half of the participants believed that they could deliver clinical services for adults with dysphagia post-stroke through telepractice as effectively as in-person service delivery. Only one participant somewhat disagreed while two participants neither disagreed nor agreed with this statement.

The majority of the participants (three participants strongly disagreeing and one somewhat disagreeing) did not believe the ninth statement that telepractice will only benefit adults with dysphagia post-stroke who live in regional towns or rural areas, not city centres whereas, one participant neither disagreed nor agreed.

The tenth statement that telepractice had the potential to improve the level of care given to adults with dysphagia post-stroke was somewhat agreed upon by one-half of the participants and one each of the participants strongly agreed, somewhat disagreed and neither disagreed nor agreed.

#### 4.3.2 Conclusion

The responses from the participants with respect to the perceptions of the use of telepractice with dysphagia post-stroke were analysed and presented in this chapter. The results started with explaining the demographics of the participants, their clinical and telepractice services, benefits, facilitators, barriers of the participants with respect to adults with dysphagia post-stroke and also enumerated the beliefs with regard to the use of telepractice as evidenced from 10 statements.

#### **Chapter Five: Discussion**

## **5.1 Introduction**

The purpose of this study was to explore the perceptions of SLTs in New Zealand on the use of telepractice with adults with dysphagia post-stroke along with their perceived benefits, facilitators, barriers, and beliefs. This chapter presents a discussion of the results of the study. The key points discussed under the perceptions of SLTs on the use of telepractice with adults with dysphagia post-stroke are as follows:

- (1) The nature of clinical services offered
- (2) Telepractice as influenced by SLT training and professional development
- (3) Telepractice as an appropriate method to deliver clinical services
- (4) Benefits and facilitators of telepractice in New Zealand
- (5) Barriers limiting the use of telepractice
- (6) Perceived benefits on the use of telepractice

#### 5.2 Clinical services offered to adults with dysphagia post-stroke via telepractice

The clinical services offered by the participants to adults with dysphagia post-stroke via telepractice varied. It is understood that effective clinical services included assessment, education for clients and caregivers, direct intervention (exercises), feeding trials, rehabilitation and compensatory strategies (muscle strength training). According to NZSTA (2012), SLTs in New Zealand used telepractice for assessment, diagnosis, case reviews and referral purposes. It is to be construed that the use of telepractice by SLTs in New Zealand additionally involves case reviews and referral purposes. The study's outcomes are in line with the findings of many authors in different countries such as in Hong Kong (Fong et al., 2020); in Australia (Hill & Miller,

2012); in India (Mohan et al., 2017), and the United States of America (Tucker, 2012). Services like consultation, follow-up sessions, family support, teacher support, natural therapy, and assessment offered by SLPs in Australia (Hill & Miller, 2012). More of paediatric population was served when compared to adult population. The present study's focus was telepractice with adults; therefore, paediatric population was not considered. As far as the use of telepractice in other countries is concerned, in the USA, the primary services provided by SLPs via telepractice were assessment, consultation and screening (Tucker, 2012); follow-up services, management and assessment in India (Mohan et al., 2017); treatment, follow-up sessions, supervision, screening and assessment in Hong Kong (Fong et al., 2020).

New Zealand uses telepractice for education for clients and caregivers. The other services such as assessment, direct intervention (exercises), feeding trials, rehabilitation and compensatory strategies (muscle strength training) are also delivered when compared to the use of telepractice in different countries as mentioned above. It could be inferred that the use of telepractice by SLTs in New Zealand is varied.

# **5.2.1** Telepractice provides SLTs with an option to expand services to underserved populations

The platforms used to deliver clinical services via telepractice play a significant role in adequate health care provision. The various media used by the participants when providing clinical services to adults with dysphagia post-stroke in New Zealand included audio calls such as phone and Skype and video conferencing. Successful telerehabilitation services for adults with stroke and the benefits of telepractice in real-time video conferencing for providing care for such individuals have been highlighted (Gregory et al., 2011). Videoconferencing is a common medium used by clinicians to provide telepractice services and online assessment of dysphagia

due to more accessibility (Hill & Miller, 2012; Lowman & Kleinert, 2017). Also, telepractice via videoconferencing with adults with dysphagia due to head and neck cancer resulted in positive client and clinician satisfaction (Ward et al., 2017). The present study's finding that videoconferencing was used as a platform for offering telepractice was supported by five out of six participants. This finding is in line with the reports of Hill and Miller (2012) and Lowman and Kleinert (2017) who stated that videoconferencing was the most effective tool for delivering clinical services via telepractice. As far as the mode of clinical service delivery via telepractice was concerned, most of the therapists in Hong Kong used the videoconferencing, telephone, and "store-and-forward" approach (Fong et al., 2020). However, none of the participants stated that they used the "store-and-forward" type of telepractice method in New Zealand.

#### 5.2.2 Methods of augmentation of clinical practice

It will be worthwhile to look into the items used to augment clinical swallow evaluation via telepractice, as is being done by SLTs in other countries. It is noted that the participants did not report different service delivery platforms. Telepractice offers diverse clinical-based applications for service delivery (Malandraki et al., 2014); the SLTs of New Zealand did not utilise these, may be due to the lack of knowledge of the use of sophisticated equipment to deliver services via telepractice. The lack of using a variety of applications could be considered a weakness and could be traced to insufficient training and professional development of SLTs. Only one-half of the SLTs in the present study planned to expand their clinical services to adults with dysphagia post-stroke via telepractice. Another outcome of the study was that two-thirds of the SLTs in the survey planned to expand their clinical services to adults with dysphagia poststroke by increasing the number of clients. The rest planned to expand their clinical services by providing communication therapy. However, none of the participants reported that they planned
to expand their clinical services by increasing the range of equipment or content of services. This indicates that the SLTs in New Zealand lack sufficient training in the use of different equipments for clinical service delivery.

Most of the respondents of a survey study in Australia wanted to expand their telepractice service for more outreach (Hill & Miller, 2012). Participants in this survey stated that they had not used any of these methods in New Zealand, such as cervical auscultation, pulse oximeter, surface electromyography or others to augment clinical swallow evaluation via telepractice. And, this is in contrast with the reports of Hill and Miller (2012); Ward et al. (2009); Ward et al. (2014), wherein items such as pulse oximeter, medical camera, web camera and extra lighting were used by SLPs in their clinical practice. However, SLTs in New Zealand are not inclined to expand their telepractice service for more outreach and broaden the client population by augmenting their practice by the increasing the range of equipments.

#### 5.2.3 Client-related factors considered when deciding to offer clinical services

Client-related factors are significant features to be considered when deciding to offer clinical service with adults with dysphagia post-stroke via telepractice. Therefore, the study participants considered various factors (as described in chapter four) when deciding to provide telepractice with adults with dysphagia post-stroke. Different factors have been suggested by earlier researchers to be considered while deciding to use telepractice with individuals with dysphagia (Malandraki et al., 2011; Morrell et al., 2017; Tinelli et al., 2017; Ward et al., 2014). The outcome of the assessment of dysphagia via telepractice depends up on the severity of the case; hence it is difficult to obtain similar agreement with that of face-to-face assessment when the severity of stroke is more. Lower levels of agreement with that of face-to-face assessment was obtained by clinicians while using telepractice with individuals with severe

stroke (Morrell et al., 2017). On the other hand, clinical trials by Malandraki et al. (2011) with 32 individuals (with a primary diagnosis of stroke or head and neck cancer) concluded that the utilization of an online telepractice system was possible for assessing and evaluating the severity of dysphagia, the degree of penetration/aspiration and for providing clinical recommendations. In the case of instrumental and non-instrumental dysphagia evaluation also, there has been a high agreement between telepractice and face-to-face delivery of service (Burns et al., 2016). The client-related factors related to clinical services with adults with dysphagia post-stroke via telepractice considered by the participants were many, comprising cognitive behavioural ability, physical, sensory ability and communication ability as far as the behaviour of the individuals is concerned. In addition, technology skills, access to hardware, software and support persons were also considered as client factors while choosing clients for clinical service delivery via telepractice by the participants. One crucial area regarding telepractice and stroke is visual impairments affecting cognition, causing adverse effects in the daily life of the individuals (Tinelli et al., 2017). However, severity of dysphagia, one of the essential factors considered by other researchers was not considered by the SLTs in New Zealand while offering telepractice. Lack of adequate professional training and development in dysphagia assessment via telepractice could be the primary reason for this. Therefore, SLTs should rectify this weakness by undergoing proper training and professional development on dysphagia management to offer clinical practices via telepractice with adults with dysphagia post-stroke.

#### 5.3 Telepractice is influenced by the training and professional development of SLTs

Training and professional development are considered essential factors for the effective use of telepractice in clinical service delivery across various health professions.

#### 5.3.1 Advantages of formal training and limitations of the curriculum in New Zealand

Formal training and professional development are essential to clinicians in handling complex cases of dysphagia via telepractice. The overall perceptions shared by the participants in professional training and development reflect that they have not undergone adequate training. The present study revealed that none of the participants had training in telepractice during their undergraduate and postgraduate course work in universities. It is to understand that the participants are constrained to use opportunity such as attending seminars, conferences; reading books and articles and peer learning to build up and improve their knowledge on telepractice to manage dysphagia. The participants had also to undergo self-directed and online learning and attend webinars on telepractice because telepractice was not included in the curriculum of universities then. There is enough opportunity in New Zealand for further improvising the training process into a more focussed one by offering hands-on training on telepractice by professional organisations. NZSTA has developed programmes by which members can continue to have professional development (NZSTA, 2018).

The importance of formal training of clinicians in handling complex cases of dysphagia via telepractice and the use of ethical guidelines reports was highlighted by many researchers (Edirippulige & Armfield, 2017; Hill & Miller, 2012; Ward et al., 2014). It is because telepractice involves a change in practice (Edirippulige & Armfield, 2017). The need for clinical training in telepractice service delivery was indicated by Miles et al. (2020). According to Mohan et al. (2017), a short-term training or a certificate course in telepractice is essential to deliver clinical services through telepractice. There is a general agreement by many researchers, as stated above, that formal training and professional development in the use of telepractice is

essential in adequate health care provision to adults with dysphagia post-stroke. Offering opportunities to gain hands-on training with different platforms of telepractice in the professional development programmes for SLTs in New Zealand is required. These would help improve the professional skills and confidence level of SLTs. They would go for more use of telepractice in managing adults with dysphagia post-stroke even though the participants expressed good confidence level in using technology in general and in clinical practice. There is an apparent need to conduct more specific and focussed training and professional development on telepractice and dysphagia management for the SLTs in New Zealand.

#### 5.3.2 Level of confidence of SLTs

One of the many factors influencing SLTs' considerations in utilising telepractice with adults with dysphagia post-stroke is confidence in using technology; in general, and in clinical practice. The significance of technology in telepractice is highlighted by many workers (Bours et al., 2009; Sharma et al., 2011; Ward et al., 2012). Having a high level of confidence, the SLTs of New Zealand could be encouraged to offer clinical services through telepractice.

### 5. 4 Benefits and facilitators of telepractice under New Zealand context

This section discusses the benefits and facilitators of the use of telepractice by SLTs in New Zealand.

# **5.4.1** Telepractice is an appropriate method to deliver clinical services with adults with dysphagia post-stroke

Telepractice is considered a vital method for delivering clinical services to adults with dysphagia post-stroke. The participants perceived numerous benefits to the use of telepractice with adults with dysphagia post-stroke at the client level, clinician level and workplace/organisation level.

#### 5.4.2 Perceived benefits of the use of telepractice in delivering clinical services

Overall, it is found that the participants suggested many benefits to the use of telepractice with adults with dysphagia post-stroke. Many of these benefits were reiterated by other researchers in various locations. Only in a few cases there was a contradiction with the statements of other researchers. The factors that are contradictory included factors such as safety precautions, privacy and legal standards, and also professional development and training of the SLTs; these are discussed later in this section.

The participants expressed that telepractice helps to remove risks associated with COVID-19 and other health factors with less exposure to hospital virus. In addition, the participants also enumerated reduced travel and costs for receiving medical care and less waiting time in a clinical setup.

A key finding arising from the study under the clinician-related benefits is that telepractice is helpful in lockdown situations and was more so because it was possible to work from home, have office sessions, and use trained therapy assistants at the client's home. The responses obtained in the study on client-related benefits are in line with the observations of several researchers (Coyle, 2012; Hill & Miller, 2012; Mashima & Doarn, 2008; McCue et al., 2010; Tucker, 2012). Many researchers (Lowe et al., 2013; Mashima & Doarn, 2008; Wales et al., 2017) stated that the quality of service offered via telepractice had similar outcomes compared with face-to-face therapy in many disorders, including dysphagia; the participants also detailed identical statements.

Good internet connectivity and availability of computers and other gadgets were considered necessary workplace-related benefits by Ayanikalath (2017); Burns et al. (2016); Malandraki et al. (2011); Ward et al. (2013). Also, privacy and legal safeguards, safety

precautions, client candidacy criteria, facilitators' role, and internet and infrastructure were suggested as workplace-related benefits of using telepractice by Gough et al. (2015) and Richmond et al. (2017). However, none of these benefits were indicated by the participants of the study. The participants of the study considered other factors as workplace/organisation-related benefits which include quality service delivery, equitable access to more clients, employee quality of life and better work-life balance.

The participants reported a range of perceived benefits to the use of telepractice concerning the client, clinician, and workplace/organisation. From the above discussion, it is clear that telepractice in general and its use in managing adults with dysphagia post-stroke has several benefits. The significant advantages concerning the client, clinician and workplace/organisation are enumerated and discussed above.

# **5.4.3** Perceived facilitators in the use of telepractice in delivering clinical services with adults with dysphagia post-stroke

Various client-related, clinician-related and workplace-related facilitators influence the delivery of clinical services via telepractice. This section discusses the role of facilitators in the use of telepractice concerning the clients, clinicians and the workplace/organisation. The part of facilitators is essential in addressing safety issues in case of emergency and also to help with completing tasks to facilitate the assessment or treatment (Malandraki et al., 2014; Sharma et al., 2011).

The frequently identified client-related facilitators by participants under the theme devices were internet support and support person. One of the important client-related facilitators identified by the participants was a trained support person, who could be a family member. The use of a trained support person as a facilitator was proposed by Burns et al. (2017); Malandraki et al. (2014); Sharma et al. (2011); Ward et al. (2013). According to Miles et al. (2020), providing the facilitators with adequate training will address safety issues that could come up during the service delivery. The facilitators should not be used as substitutes for specialists for dysphagia. It is to be noted that facilitators should only be assisting the client.

In India, a survey study found that clinicians learned to use telepractice through personal experience than formal training (Mohan et al., 2017). Similarly in the present survey, it has been found that participants have undergone self-directed study with respect to telepractice with dysphagia management. Prior training and preparation are essential for accessing more complex cases of dysphagia in clients while using telepractice (Edirippulige & Armfield, 2017; Hill & Miller, 2012; Ward & Burns, 2014). To conclude, the participants were attributing more importance to training and experience with technology as client-related facilitators.

The most frequently identified workplace/organisation-related facilitator by the participants was work from home. Appropriate space availability at the client's home and clinician facility has been identified as an essential workplace-related facilitator by Burgoyne and Cohn (2020). It is necessary for practical work from home, such as delivery and reception of services without distractions, as identified by the SLTs of New Zealand. The other workplace-related facilitator includes support by the organisation because telepractice supports the organisation. Factors such as safety precautions, privacy and legal standards, and internet specification in telepractice for dysphagia management were emphasised by Gough et al. (2015) and Richmond et al. (2017). The participants did not identify these significant factors as facilitators. Also, there is a general agreement that formal training in telepractice is necessary in effective health care provision to adults with dysphagia post-stroke. To conclude, it seen that the participants were attributing due significance to work setting and support by an organisation

concerning workplace/organisation-related facilitators of using telepractice for adults with dysphagia post-stroke.

In general, it is clear from the above discussion that telepractice with adults with dysphagia post-stroke has a variety of facilitators as perceived by the participants. Having an excellent organisational set up was considered a crucial factor in facilitating telepractice, according to Ward et al. (2013). In the New Zealand context, training and experience with technology and a good organisation set up with adequate support from the organisation are the major factors influencing facilitators in the use of telepractice with dysphagia post-stroke. To conclude, it is imperative to access all the factors mentioned above to ensure the smooth delivery of clinical services via telepractice.

#### 5.5 Barriers limiting the use of telepractice in New Zealand

Telepractice has immense potential in providing a better quality of life to adults with dysphagia post-stroke under different systems of health care in many contexts. Still, one should understand that implementing telepractice has numerous barriers. This section discusses the obstacles of clinical services in the use of telepractice with adults with dysphagia post-stroke concerning the client, clinician and the workplace/organisation.

The participants' most frequently identified client-related barrier was poor internet quality, which was suggested by most participants. The significance of technology in telepractice is highlighted by many researchers (Bours et al., 2009; Sharma et al., 2011; Ward et al., 2012). While using telepractice, the images should be of high quality for better assessment of people with dysphagia, and that bandwidth plays an essential role, according to Sharma et al. (2011). On the other hand, Ward et al. (2012) reported a need for a high-quality lighting source, digital stethoscope, and good quality microphones as factors of importance. Client character was the other major factor identified by the participants concerning client-related barriers in the use of telepractice. Different workers identified several factors concerning client suitability:

- case-to-case basis client appropriateness (Brennan et al., 2010).
- Factors like hearing impairment and movement disorders (Ward et al., 2012).

• The medical diagnosis of the clients and associated co-morbidities (Ward et al., 2014). In addition, Miles et al. (2020) came up with the idea of client safety as a significant client-based barrier in the use of telepractice and Malandraki et al. (2021) stated reimbursement hurdle as a client-based barrier in the use of telepractice. However, the latter two factors were not identified as factors of importance by the participants.

The significant clinician-related barriers identified by the participants were:

- Fear of technology and, dislike of change.
- Preference for an in-person meeting.
- Lack of willingness to change standard practice.

Also, the lack of effort and skills required to support clients with technology issues was considered a barrier. These responses are in line with the statements of May and Erickson (2014) and Miles et al. (2020), who reiterated that the attitudes and perceptions of the clinicians play a significant role in the use of telepractice. The difficulty to convey empathy via telepractice when working with clients with linguistically or culturally diverse backgrounds, as stated by Pfitzner et al. (2020), was not identified as a barrier by the participants. It could conclude that variations exist in identifying client-related barriers by the participants when compared with those of other researchers. Special mention should be made of factors such as preference for an in-person meeting and lack of willingness to change standard practice. These factors could be perceived as

the major stumbling blocks limiting the increased use of telepractice by the SLTs in New Zealand, and efforts should be made to alleviate these barriers.

Lack of access to a quiet clinic room and lack of space to work in a quiet room were considered significant workplace/organisation-related barriers in using telepractice. The response that requirement of a quiet room with sufficient space is to be considered significant for telepractice to be effective. Similar observation was stated by Bours et al. (2009) which indicate that it is not possible to have telepractice in all clinical settings.

Also, inexpensive web-based software with limited privacy and security settings is a barrier. The platforms used for telepractice should be made secure in terms of patient confidentiality and data storage (Doss et al., 2018; Emezue, 2020). Therefore, the response by the participants that lack access to a quiet clinic room and lack of space to work in a quiet room as workplace/organisation-related barriers of using telepractice is justified.

Lack of shared laptops and poor internet quality were suggested as other workplacerelated barriers. These have been endorsed by a survey study in Australia, which found that the commonly reported barriers by SLPs while providing telepractice services included internet connectivity problems, limited access to technology to conduct services online and lack of assessment and treatment resources (Hill & Miller, 2012). In addition, Miles et al. (2020) and Richmond et al. (2017) also suggested that internet connectivity and technology specifications act as barriers in implementing telepractice in the management of patients with dysphagia. One SLT in New Zealand expressed inability to use telepractice because objective measurements such as cough testing cannot be assessed via telepractice. The above response by one participant that objective measurements could not be assessed via telepractice could have stemmed from the fact that the SLTs in New Zealand lack sufficient training and professional development in the

use of telepractice. It is found that the uptake of telepractice has increased among the SLTs in countries like India and Brazil, following the COVID-19 pandemic (Aggarwal et al., 2020; Rech et al., 2020). Therefore, clients and clinicians expressed more satisfaction in telepractice (Brennan et al., 2009) and the use of telepractice in the remote assessment of swallowing disorders is considered valid and feasible as per Sharma et al. (2011).

From the above discussion, it is clear that telepractice in general and its use in the management of adults with dysphagia post-stroke encompass many barriers and limitations, even though reports state that it is a better system for addressing health care. The participants reported a range of perceived barriers concerning the client, clinician, and workplace/organisation. The significant barriers concerning the client, clinician and workplace/organisation were enumerated and discussed above.

#### 5.6 Beliefs of SLTs in the use of telepractice with adults with dysphagia post-stroke

This section discusses the beliefs of SLTs on the use of telepractice. A key finding arising from the study was that SLTs while performing telepractice with adults with dysphagia poststroke are not equipped with sufficient equipment, training, and professional development to augment their clinical service delivery.

#### **5.6.1** Telepractice is an appropriate method to deliver clinical services

The primary objective of using telepractice is to improve access to health care services by providing clinical services from a distance. Most participants viewed telepractice as an appropriate method to deliver SLT services to adults with dysphagia post-stroke. This is in agreement with the observations of several authors that telepractice is an effective service delivery platform compared to face-to-face therapy (Regina Molini-Avejonas et al., 2015; Sharma et al., 2013; Swales et al., 2020). Only a few SLTs considered telepractice to be less

effective than face-to-face therapy (Fong et al., 2020). This consideration might have stemmed from the fact that SLTs might have certain misconceptions about the effectiveness of telepractice. The absence of professional development trainings for the SLTs on the use of telepractice could have added to this thought.

The statement telepractice will eventually become a mainstream delivery method for SLT services for adults with dysphagia post-stroke had a mixed response with half of the participants somewhat disagreeing, one participant neither disagreeing nor agreeing, and two participants strongly agreeing.

Furthermore, there is an increased uptake of telepractice among the SLTs in countries like India and Brazil following the COVID-19 pandemic (Aggarwal et al., 2020; Rech et al., 2020). Therefore, it is logical to perceive that telepractice will become a mainstream service delivery method for SLT for adults with dysphagia post-stroke, even though one-half of the participants somewhat disagreed with the statement. Therefore, it should be because telepractice helps provide clinical services to adults with dysphagia post-stroke irrespective of the location.

Telepractice had the potential to improve the level of care given to adults with dysphagia post-stroke was strongly agreed upon by one-half of the participants. This observation is in line with the observations of earlier researchers that improved quality of service in the client's functional environment could be provided by telepractice (McCue et al., 2010) and that many SLPs in Australia wanted to expand their telepractice service for more outreach and to broaden the client population (Hill & Miller, 2012) . As such, it could be concluded that there is a general agreement that telepractice had the potential to improve the level of care given to adults with dysphagia post-stroke. This could possibly be due to improve daccess to specialists with

expertise in dysphagia and more frequent service because of enhanced access, as suggested by Tucker (2012).

#### 5.6.2 Access to telepractice by adults with dysphagia post-stroke

Access to telepractice is an essential factor in delivering clinical services. There is more agreement with the statement that in 10 years, most adults in New Zealand with dysphagia poststroke will have access to clinical services via telepractice. Given the fact that improved access to clinical services for remotely located families and the provision of health care continuously to a client can be achieved by telepractice (Burgoyne & Cohn, 2020; Cole et al., 2019), it could perceive that in 10 years, most adults in New Zealand with dysphagia post-stroke will have access to clinical services via telepractice. Furthermore, assuming a generally similar trend with the elderly population in New Zealand, it could perceive that over the next ten years, older clients' limited familiarity and interest in technology will become less of a barrier in the use of telepractice.

The extent of area that could be covered by SLTs to offer clinical services is found to be vast. In addition, telepractice enabled clinicians to cover a large geographical area and provide more clinical services to needy people (Mashima & Doarn, 2008). However, many reports are stating that telepractice improved access to SLT services for underserved populations (Burgoyne & Cohn, 2020; Cole et al., 2019; Houn & Trottier, 2003). In this connection it is apt to explore the awareness of stroke in an ethnic urban population. It was found that participants of a survey from Pacific ethnic groups in New Zealand were not able to identify the risks associated with stroke when compared to New Zealand/European participants Bay et al. (2015). Also, those participants from Maori, Pacific and Asian origins were less likely to go in for immediate medical attention.

The statement that over the next 10 years, older clients' limited familiarity and interest in technology will become less of a barrier to telepractice had a mixed response from the participants. In a survey study in Australia with adults with Parkinson's disease, Swales et al. (2020) suggested that SLPs perceived that older individuals' limited familiarity and interest with technology is set to reduce over a period of 10 years.

#### 5.6.3 Client-clinician rapport via telepractice

Developing a good rapport between the client and the clinician is essential in delivering clinical services via telepractice. The majority of the participants (five out of six) somewhat disagreed with the statement that they could not develop the same level of client-clinician rapport via telepractice as is achieved in person. Only one participant strongly agreed with this statement. The fact that SLTs could create the exact status of client-clinician connection via telepractice has been endorsed by earlier researchers. According to Sharma et al. (2013), the majority of the clinicians felt that they developed a good rapport with their clients while providing clinical service for dysphagia via telepractice. A good level of caregiver satisfaction with clinician empathy and rapport in the use of telepractice for clinical service delivery was mentioned by (Orlando et al., 2019).

#### **5.6.4** Workplace set up and access to telepractice

Workplace set up is a significant consideration in the use of telepractice delivery. Two of the participants neither disagreed nor agreed and four out of six somewhat agreed or strongly agreed with the statement that their workplace was open. They were interested in using telepractice to provide clinical services for adults with dysphagia post-stroke. It states that organisational access to appropriate technology and information technology support for practicing telepractice is a significant corporate benefit of the use of telepractice (Pfitzner et al.,

2020). In addition, it suggests that growing evidence projects the advantages of telepractice over the traditional in-person approach regarding service efficiency, better quality service outcomes, and seeing more clients promptly (Regina Molini-Avejonas et al., 2015). These could be considered the significant reasons for the participants' interest in using telepractice to provide clinical services for adults with dysphagia post-stroke.

#### 5.6.5 Telepractice vs face-to-face clinical service delivery

There is a general debate among the SLTs whether telepractice or face-to-face delivery of clinical services is better or on par with each other. One-half of the participants believed that it could deliver clinical services for adults with dysphagia post-stroke through telepractice as effectively as in-person service delivery. One participant somewhat agreed to this statement, and two participants strongly agreed. There is a general agreement with many SLTs that it could deliver clinical services through telepractice as effectively as in-person service delivery in many disorders, including dysphagia (Lowe et al., 2013; Mashima & Doarn, 2008; Regina Molini-Avejonas et al., 2015; Sharma et al., 2013; Wales et al., 2017). Also, in some cases, telepractice could be considered superior to face-to-face delivery of clinical services in terms of better client satisfaction and service delivery (Burns et al., 2017). The majority of the respondents in a survey study in Australia believed that clinical services could be delivered via telepractice to individuals with Parkinson's disease as effectively as in face-to-face therapy and that telepractice could become a mainstream service delivery option (Swales et al., 2020). Also, it states that evidence to date supports the use of telepractice as an equivalent alternative to in-person dysphagia care for adults in inpatient and outpatient settings (Malandraki et al., 2021). Thus, it could perceive that telepractice is an appropriate clinical service delivery method for adults with dysphagia poststroke and is as effective as in-person service delivery.

To the statement that SLTs would use telepractice only if inpatient service delivery is not an option, one participant strongly disagreed, and three participants somewhat disagreed. Only one of the participants somewhat agreed and strongly agreed with the statement. According to Swales et al. (2020), a similar view might have come from the primary purpose of the remote delivery model, which is to reach those in underserved regions and who cannot easily access clinical support. As there is a general disagreement by the participants, it could perceive that SLTs of New Zealand do not comply with the statement that they would use telepractice only if inpatient service delivery is not an option.

To conclude, participants considered telepractice as a feasible and effective clinical service delivery option. The agreement could be generalised by responding to the 10 statements provided by the participants. It may be construed that the participants viewed telepractice as a viable clinical service delivery option that may provide improved health care and benefits to adults with dysphagia post-stroke, irrespective of the location/region.

#### **Chapter Six: Conclusion**

This chapter summarises the purpose, rationale and key findings of the study which aimed to explore the perceptions of speech-language therapists (SLTs) in New Zealand on the use of telepractice with adults with dysphagia post-stroke along with perceived benefits, facilitators, barriers, and beliefs. The strengths and limitations of the study are discussed. The implications for clinicians, client educators and vocational training and the thoughts on future research are also enumerated.

#### 6.1 Purpose, design and rationale

A review of the literature on the use of telepractice by SLTs with adults with dysphagia post-stroke reveals a lack of research in this field of clinical practice in New Zealand. It could be because telepractice with adults with dysphagia post-stroke is a highly specialised technical field. As a result, only significantly fewer SLTs in New Zealand may be using this clinical service delivery method. However, the work done elsewhere suggests that this service delivery model can be effective and efficient with a lot of benefits and good client and clinical satisfaction.

The online survey participants were SLTs working in New Zealand with experience in using telepractice with adults with dysphagia post-stroke. This study has helped fill the literature gap by bringing out the perceptions of SLTs in New Zealand and describe the outcomes of the use of telepractice with adults with dysphagia post-stroke. The findings also would contribute to the growing body of information on the use of telepractice in dysphagia with adults, including diagnosis and interventions.

The study used a descriptive cross-sectional survey design. Predominantly quantitative data and a few qualitative parameters have been collected via the survey platform Qualtrics to describe the perceptions and beliefs of the SLTs who use telepractice. The participants were

recruited indirectly through the New Zealand Speech-Language Therapists' Association and the Massey University Speech and Language Programme Facebook page.

The demographics of the respondents, their clinical settings, their confidence with the technology, different clinical services along with the platforms used for delivering telepractice, the benefits, facilitators and barriers on the use of telepractice concerning the clients, clinicians and workplace/organisation were studied and enumerated. Finally, the beliefs of the SLTs who use telepractice with adults with dysphagia post-stroke, which was captured using ten statements on a five-point scale, are discussed.

The study results suggested that there are many benefits and positive effects on the use of telepractice with adults with dysphagia post-stroke. Also, the perceptions of the SLTs concerning clients, clinicians and the workplace varied. In addition, the study deliberates the practical issues that need considering while working with adults with dysphagia post-stroke via telepractice.

#### 6.2 Limitations of the study

The results of this study should be considered against the limitations. The limitations include:

- (1) Sample size
- (2) COVID-19 lockdown
- (3) Regional representation
- (4) Lack of experience with ethnic group population
- (5) Limited field of study
- (6) Confidential nature of data collection
- (7) Type and time of receiving training on telepractice by SLTs

The number of participants (n=6) can be construed as a limitation; despite sending a follow-up reminder asking the organisation to remind the SLTs who met the inclusion criteria for

participation in the survey. The reduced number of participants should be considered in the light of the fact that the study focussed on the use of telepractice with adults with dysphagia poststroke, which is a super-speciality in medicine.

There may not be many SLTs who have a similar population as their clients; this could be one reason for the fewer participants. Also, when compared to other countries, the lockdown period due to the COVID-19 pandemic was short in New Zealand at the time of the study; this could mean that telepractice may not have been as widespread as is the case elsewhere. When we look at these facts, we should consider that this number of participants is adequate for eliciting information on the use of telepractice. Although the number of participants involved in the study was less, it is representative of the SLTs working in the super-specialised field of adults with dysphagia post-stroke.

The SLTs from only four regions (out of 16 regions) in New Zealand have participated in the study. The responses of the participants were based on the region where they are working. The same should be considered as a limitation of the study as there could be a bias as far as the regions are concerned, and the outcomes could not be generalised throughout New Zealand. In addition, the participants have not mentioned their experience in using telepractice with Māori and Pacific people population; this is to be considered as another limitation. It could be because the focus of the study was not on Māori and Pacific people. The participants might not have had individuals from Māori and Pacific people population as their clients. As a result, the study could not assess the participants' perceptions in offering clinical services to diverse ethnic groups in New Zealand.

The study focussed only on adults with dysphagia post-stroke. Any other population is not considered (like paediatric and people with other neurological problems) and this could be interpreted as a limitation of the study. Furthermore, it did not include the age group of the participants in the questionnaire to maintain privacy and confidentiality and is a limitation because it is impossible to ascertain whether the perceptions and beliefs have come from participants of younger or older age groups. Also, it could not analyse the differences in the perception between the age groups of the participants. Finally, even though the study elicited the information at the stage at which the participants received training in telepractice, there was no question detailing the type and appropriate time of training to be provided to SLTs on telepractice and this forms another limitation of the present study.

#### 6.3 Strengths of the study

The study's strength lies in the development of the questionnaire in that this could measure the parameters collected in the true sense. The results showed agreement with the outcomes obtained by other researchers in different locations, even though many factors affect the effectiveness of telepractice. The participants identified a range of benefits to using telepractice and suggested several facilitators for further improvement. In addition, it also delineated significant barriers to the use of telepractice.

The use of quantitative and a few qualitative questions to elicit responses from the participants resulted in improving the comprehensiveness of the overall findings and expanded the dimensions of the study. The understanding of the perceptions of SLTs on the use of telepractice with adults with dysphagia post-stroke has helped to outline solutions to improve the use of this service delivery model to this client population. This is in the light of the fact that there is no comprehensive data available on the perceptions of SLTs in the use of telepractice in New Zealand.

#### 6.4 Validity and reliability of the study

Steps were taken to see that the respondents understood all the survey questions as intended. After the pilot study removed some questions and a few modifications, this study did a little reframing for six questions, including sub-titles such as client-related, clinician-related, and workplace/organisation-related factors. Further, the questionnaire covered a range of topics related to the use of telepractice. A proper representation of the total population has been surveyed even though the number of participants was small, as the study pertained to a superspeciality clinical practice, thus assuring reliability. Also, the results obtained are consistent and add to the reliability of the study. It could be suggested that this study has helped to understand the practical aspects of telepractice use. The study also has helped in bringing out the usefulness of this service delivery as an effective and efficient model with a lot of benefits to the clinician as well as the clients, even though with some limitations and barriers. Thus, it could be stated that the methodology of the study has helped to bring out the factors affecting clinician acceptance of telepractice, which are significant to the uptake and expansion of telepractice among the SLTs in New Zealand. These key factors must be addressed to facilitate the improved use and sustainability of telepractice in clinical services in New Zealand.

#### 6.5 Implications for clinical practice, client education, and training

The significant outcomes of the study have the potential to be extrapolated and adopted into clinical trials by the SLTs working with adults with dysphagia post-stroke. The data on the perceived benefits, facilitators, barriers and beliefs on the use of telepractice by the participants will help in understanding the intricacies in the implementation of telepractice in clinical service delivery. These would also help in formulating vocational training programmes and imparting client education for easy adoption of telepractice. Even though the study has certain limitations, many factors form the basis for the widespread use of telepractice by SLTs in New Zealand. It also reflects the opinions expressed by the participants. It considers that the study provides a strong base for telepractice and, therefore, can result in greater uptake by the SLTs for managing adults with dysphagia post-stroke. While the focus of the study is on the use of telepractice with adults with dysphagia post-stroke, the results could be generalised into the delivery of clinical services in other primary health conditions. And, this should be considered in the light of the COVID-19 pandemic when the utility and safety of traditional face-to-face delivery of clinical services was highly variable and guarded across many countries.

The following are the implications for areas of clinical practice, client education and training derived from the study:

(1) Use of improved devices for offering telepractice and equipments for assessing and managing dysphagia via telepractice.

(2) Formulating vocational training programmes for imparting hands-on training to clinicians and educating clients and caretakers for easy adoption of telepractice.

(3) Research on best practice models to provide the opportunity for telepractice to be integrated fully into the health care system in New Zealand.

#### **6.6 Directions for future research**

The study results have revealed the perceptions of SLTs in New Zealand on the use of telepractice with adults with dysphagia post-stroke. These, included details on the benefits, facilitators, and barriers of the SLTs regarding clients, clinicians, the workplace, and their beliefs. The enumerated obstacles call for further studies of these aspects, as these are the constraints faced by the SLTs in New Zealand, and these require to be addressed. The study's outcomes add to the knowledge bank in the field of telepractice with adults with dysphagia post-

stroke, as this is the first of the kind study taken up in New Zealand. In addition, the results are helpful for SLTs for furthering their knowledge on several aspects of telepractice use; these are emulated by more SLTs from different regions of New Zealand. The results, thus, could inspire other researchers to carry out further studies on the multidimensional aspects of telepractice use not only in the management of dysphagia but also in other health disorders with different client populations. Many areas need further research, such as the systems for offering telepractice and the augmentation of the same for expanding telepractice service. Further research in this field will provide the opportunity for SLTs to use telepractice and to integrate this system more into the health care provision in New Zealand.

#### **6.7** Conclusion

This chapter emphasises the concluding remarks of the study. The concluding remarks have been presented under different topics: the purpose, design and rationale of the study; limitations of the study; strengths of the study; validity and reliability of the study; implications for clinical practice, client education, and training; and directions for future research. Overall, the study revealed the perceptions of the SLTs in New Zealand in using telepractice with adults with dysphagia post-stroke in reference to the benefits, facilitators, barriers and beliefs. These lead to the fact that telepractice could be considered a vital method for delivering clinical services to adults with dysphagia post-stroke and that the participants would like to use telepractice. Also, it is understood that the participants are of the opinion that telepractice is on par with face-to-face delivery of clinical services. The perception that telepractice would help to avoid risks withCOVID-19 is to be construed as an important and practical reason for the increased interest in the use of this service delivery option. Given a chance, the participants would like to augment their clinical set up with the addition of more clients. It is also imperative

that appropriate training be provided to the SLTs in New Zealand in the use of telepractice. The participants believed that telepractice resulted in healthier and happier employees providing a better work-life balance and general quality of life. This belief is an important step in the increased acceptance of using this method of clinical service delivery by the SLTs of New Zealand, even though there are limitations.

#### REFERENCES

- Aggarwal, K., Patel, R., & Ravi, R. (2020). Uptake of telepractice among speech-language therapists following COVID-19 pandemic in India. *Speech, Language and Hearing*, 1-7.
- Akemoglu, Y., Muharib, R., & Meadan, H. (2020). A systematic and quality review of parentimplemented language and communication interventions conducted via telepractice. *Journal of Behavioral Education*, 29(2), 282-316.
- Albaum, G. S., & Smith, S. M. (2006). Handbook of marketing research, London. Sage.
- Ary, D., Jacobs, L. C., Irvine, C. K. S., & Walker, D. (2018). *Introduction to research in education*. Cengage Learning.
- ASHA. (2019). The national outcomes measurement system: Pre-kindergarten national data report. *National Center for Evidence-Based, Practice in Communication Disorders*(p1).
- Ayanikalath, S. N. (2017). Investigation of the effectiveness of an ecologically valid telerehabilitation system for the assessment and primary management of neurogenic dysphagia in a resource constrained country [Doctoral Thesis,
- Bay, J. L., Spiroski, A.-M., Fogg-Rogers, L., McCann, C. M., Faull, R. L., & Barber, P. A. (2015). Stroke awareness and knowledge in an urban New Zealand population. *Journal of Stroke and Cerebrovascular Diseases*, 24(6), 1153-1162.
- Bedford, J., Enria, D., Giesecke, J., Heymann, D. L., Ihekweazu, C., Kobinger, G., . . . Schuchat, A. (2020). COVID-19: Towards controlling of a pandemic. *The lancet*, 395(10229), 1015-1018.
- Bours, G., Speyer R, Lemmens J, Limburg M, & D., W. (2009). Bedside screening tests vs. videofluoroscopy or fibreoptic endoscopic evaluation of swallowing to detect dysphagia in patients with neurological disorders: Systematic review. *Journal of Advance Nursing*, 65(3), 477-493.
- Brennan, D., Tindall, L., Theodoros, D., Brown, J., Campbell, M., & Christina, D. (2010). A blueprint for telerehabilitation guidelines. *Telemedicine and E-Health*, *17*, 662-665.
- Brennan, D. M., Mawson, S., & Brownsell, S. (2009). Telerehabilitation: Enabling the remote delivery of healthcare, rehabilitation, and self management. In *Advanced Technologies in Rehabilitation* (pp. 231-248). IOS Press.
- Bryman, A. (2016). Social research methods. Oxford university press.
- Burgoyne, N., & Cohn, A. S. (2020). Lessons from the transition to relational teletherapy during COVID-19. *Family Process*, *59*(3), 974-988.
- Burns, C. L., Ward, E. C., Hill, A. J., Kularatna, S., Byrnes, J., & Kenny, L. M. (2017). Randomized controlled trial of a multisite speech pathology telepractice service providing swallowing and communication intervention to patients with head and neck cancer: Evaluation of service outcomes. *Head and Neck*, 39(5), 932-939.
- Burns, C. L., Ward, E. C., Hill, A. J., Phillips, N., & Porter, L. (2016). Conducting real-time videofluoroscopic swallow study via telepractice: A preliminary feasibility and reliability study. *Dysphagia*, 31(3), 473-483.
- Burns, N., & Grove, S. K. (2005). *The practice of nursing research: Conduct, crituique and utilization*. Elsevier.
- Carey, B., O'Brian, S., Onslow, M., & al, e. (2010). Randomized controlled non-inferiority trial of a telehealth treatment for chronic stuttering: The camperdown program. *International Journal of Language and Communication Disorders*, *45*, 108-120.

- Cason, J., & Cohn, E. R. (2014). Telepractice: An overview and best practices. *Perspectives on Augmentative and Alternative Communication*, 23(1), 4-17.
- Cassel, S. G. (2016). Trial dysphagia interventions conducted via telehealth. *International Journal of Telerehabilitation*, 8(2), 71-76.
- Cherney, L. R., Kaye, R. C., & Hitch, R. S. (2011). The best of both worlds: Combining synchronous and asynchronous telepractice in the treatment of aphasia. *Perspectives on Neurophysiology and Neurogenic Speech and Language Disorders*, 21(3), 83-93.
- Cole, B., Pickard, K., & Stredler-Brown, A. (2019). Report on the use of telehealth in early intervention in Colorado: Strengths and challenges with telehealth as a service delivery method. *International Journal of Telerehabilitation*, 11(1), 33.
- Combi, C., Pozzani, G., & Pozzi, G. (2016). Telemedicine for developing countries. *Applied Clinical Informatics*, 7(04), 1025-1050.
- Coyle, J. (2012). Tele-dysphagia management: An opportunity for prevention, cost-savings and advanced training. *International Journal of Telerehabilitation*, 4(1), 41-46. https://doi.org/https://doi.org/10.5195/ijt.2012.6093
- Creswell, J., & Guetterman, T. (2019). Educational research: Planning, conducting, and evaluating quantitative and qualitative research. *Pearson, 6th edn Boston*.
- CSULB. (2020). What is Qualtrics? https://csulb.libguides.com/qualtrics.
- Dillman, D. A., Smyth, J. D., & Christian, L. M. (2014). Internet, phone, mail and mixed mode surveys: The tailored design method. *4th ed*.
- Doss, B. D., Feinberg, L. K., Rothman, K., Roddy, M. K., & Comer, J. S. (2018). Using technology to enhance and expand interventions for couples and families: Conceptual and methodological considerations. *Journal of Family Psychology*, 31(8), 983.
- Edirippulige, S., & Armfield, N. (2017). Education and training to support the use of clinical telehealth: A review of the literature. *Journal of Telemedicine and Telecare*, 23(2), 273-282.
- Emezue, C. (2020). Digital or digitally delivered responses to domestic and intimate partner violence during COVID-19. *JMIR Public Health and Surveillance*, *6*(3), e19831.
- Faux, S. G., Eagar, K., Cameron, I. D., & Poulos, C. J. (2020). COVID-19: Planning for the aftermath to manage the aftershocks. *The Medical Journal of Australia*, *213*(2), 60-61. e61.
- Fong, B., Fong, A. C. M., & Li, C. K. (2011). *Telemedicine technologies: Information technologies in medicine and telehealth*. John Wiley & Sons.
- Fong, R., Tsai, C. F., & Yiu, O. Y. (2020). The implementation of telepractice in speech language pathology in Hong Kong during the COVID-19 pandemic. *Telemedicine and E-Health*, 27(1), 30-38.
- Gibbs, G. R. (2015). Analyzing qualitative data (Vol. 6). Sage.
- Gough, F., Budhrani, S., Cohn, E., Dappen, A., Leenknecht, C., Lewis, B., . . . Roberts, L. (2015). ATA practice guidelines for live, on-demand primary and urgent care. *Telemedicine and E-Health*, 21(3), 233-241.
- Gregory, P., Alexander, J., & Satinsky, J. (2011). Clinical telerehabilitation: Applications for physiatrists. *3*(7), 647-656.
- Grillo, E. U. (2019). Building a successful voice telepractice program. *Perspectives of the ASHA Special Interest Groups*, *4*(1), 100-110.

- Grogan-Johnson, S., Alvares, R., Rowan, L., & Creaghead, N. (2010). A pilot study comparing the effectiveness of speech language therapy provided by telemedicine with conventional on-site therapy. *Journal of Telemedicine and Telecare*, *16*(3), 134-139.
- Hill, A. J., & Miller, L. E. (2012). A survey of the clinical use of telehealth in speech-language pathology across Australia. *Journal of Clinical Practice in Speech-Language Pathology*, 14(3), 110-117.
- Houn, B., & Trottier, K. (2003). Meeting the challenge of rural service delivery. *The ASHA Leader*, 8(21), 2-15.
- Irwin, D. L., Lass, N. J., Pannbacker, M., Koay, M. E. T., & Whited, J. S. (2019). *Clinical* research methods in speech-language pathology and audiology. Plural Publishing.
- Kantarcigil, C., & Malandraki, G. A. (2017). First step in telehealth assessment: A randomized controlled trial to investigate the effectiveness of an electronic case history form for dysphagia. *Dysphagia*, *32*(4), 548-558.
- Keck, C. S., & Doarn, C. R. (2014). Telehealth technology applications in speech-language pathology. *Telemedicine and E-Health*, 20(7), 653-659.
- Klein, A. (2020). New Zealand's success.
- Lowe, R., O'brian, S., & Onslow, M. (2013). Review of telehealth stuttering management. *Folia Phoniatrica et Logopaedica*, 65(5), 223-238.
- Lowman, J. J., & Kleinert, H. L. (2017). Adoption of telepractice for speech-language services: A statewide perspective. *Rural Special Education Quarterly*, *36*(2), 92-100.
- Maggio, M. G., De Luca, R., Manuli, A., & Calabrò, R. S. (2020). The five 'W' of cognitive telerehabilitation in the COVID-19 era.
- Malandraki, G. A., Arkenberg, R. H., Mitchell, S. S., & Malandraki, J. B. (2021). Telehealth for dysphagia across the life span: Using contemporary evidence and expertise to guide clinical practice during and after covid-19. *American Journal of Speech-Language Pathology*, 30(2), 532-550.
- Malandraki, G. A., Markaki, V., Georgopoulos, V. C., Bauer, J. L., Kalogeropoulos, I., & Nanas, S. (2013). An international pilot study of asynchronous teleconsultation for oropharyngeal dysphagia. *Journal of Telemedicine and Telecare*, 19(2), 75-79.
- Malandraki, G. A., McCullough, G., McWeeny, E., X., H., & Perlman, A., 1485-1496. (2011). Teledynamic evaluation of oropharyngeal swallowing. *Journal of Speech, Language, and Hearing Research*, 54, 1485-1496.
- Malandraki, G. A., Roth, M., & Sheppard, J. J. (2014). Telepractice for pediatric dysphagia: A case study. *International Journal of Telerehabilitation*, 6(1), 3.
- Mashima, P. A., & Doarn, C. R. (2008). Overview of telehealth activities in speech-language pathology. *Telemedicine and E-Health*, 14(10), 1101-1117.
- Maxwell, D. L., & Satake, E. (2006). *Research and statistical methods in communication sciences and disorders*. Thomson/Delmar Learning.
- May, J., & Erickson, S. (2014). Telehealth: Why not? *Journal of Clinical Practice in Speech-Language Pathology*, *16*(3), 147-151.
- McCue, M., Fairman, A., & Pramuka, M. (2010). Enhancing quality of life through telerehabilitation. *Physical Medicine and Rehabilitation Clinics*, 21(1), 195-205.
- Miles, A., Connor, N. P., Desai, R. V., Jadcherla, S., Allen, J., Brodsky, M., . . . Moss, M. (2020). Dysphagia care across the continuum: A multidisciplinary dysphagia research society taskforce report of service-delivery during the covid-19 global pandemic. *Dysphagia*, 36(2), 170-182.

- Miles, A., Connor, N. P., Desai, R. V., Jadcherla, S., Allen, J., Brodsky, M., . . . Moss, M. (2021). Dysphagia care across the continuum: A multidisciplinary dysphagia research society taskforce report of service-delivery during the covid-19 global pandemic. *Dysphagia*, 36(2), 170-182.
- Mohan, H. S., Anjum, A., & Rao, P. K. (2017). A survey of telepractice in speech-language pathology and audiology in India. *International Journal of Telerehabilitation*, 9(2), 69-80.
- Morrell, K., Hyers, M., Stuchiner, T., Lucas, L., Schwartz, K., Mako, J., . . . Yanase, L. (2017). Telehealth stroke dysphagia evaluation is safe and effective. *Cerebrovascular Diseases*, 44(3-4), 225-231.
- Nordio, S., Innocenti, T., Agostini, M., Meneghello, F., & Battel, I. (2018). The efficacy of telerehabilitation in dysphagia patients: A systematic review. *Acta Otorhinolaryngologica Italica*, *38*(2), 79-85.
- NZSTA. (2012). Scope of practice. (https://speechtherapy.org.nz/wpcontent/uploads/2013/09/NZSTA-Scope-of-Practice-2012.pdf)
- NZSTA. (2018). New Zealand Speech-Language Therapists' Association. https://speechtherapy.org.nz.
- Orlando, J. F., Beard, M., & Kumar, S. (2019). Systematic review of patient and caregivers' satisfaction with telehealth videoconferencing as a mode of service delivery in managing patients' health. *PLOS ONE*, *14*(8), e0221848.
- Pfitzner, N., Fitz-Gibbon, K., McGowan, J., & True, J. (2020). When home becomes the workplace: Family violence, practitioner wellbeing and remote service delivery during COVID-19 restrictions Monash University].
- Piron, L., Turolla, A., Tonin, P., Piccione, F., Lain, L., & Dam, M. (2008). Satisfaction with care in post-stroke patients undergoing a telerehabilitation programme at home. *Journal of Telemedicine and Telecare*, 14(5), 257-260.
- Punch, K. F. (2013). Introduction to social research: Quantitative and qualitative approaches. sage.
- Rech, R. S., Bulgarelli, P. T., Condessa, A. M., Santos, C. M. d., Hilgert, J. B., & Goulart, B. N.
  G. d. (2020). Access and use of speech-language therapy services in Porto Alegre, Brazil: A population-based study. *Ciencia and Saude Coletiva*, 25, 817-825.
- Regina Molini-Avejonas, D., Rondon-Melo, S., de La Higuera Amato, C. A., & Samelli, A. G. (2015). A systematic review of the use of telehealth in speech, language and hearing sciences. *Journal of Telemedicine and Telecare*, 21(7), 367-376.
- Reynolds, A. L., Vick, J. L., & Haak, N. J. (2009). Telehealth applications in speech-language pathology: A modified narrative review. *Journal of Telemedicine and Telecare*, 15(6), 310-316.
- Richmond, T., Peterson, C., T., C., & al., e. (2017). American telemedicine associations's principles of delivering telerehabilitation services. *International Journal of Telerahabilitation*, *9*(2), 63-68.
- Rogante, M., Grigioni, M., Cordella, D., & Giacomozzi, C. (2010). Ten years of telerehabilitation: A literature overview of technologies and clinical applications. *NeuroRehabilitation*, 27, 287-304.
- Rogers, B. (2020). A practical guide to working with children and families through telehealth. In: Adelaide: Emerging Minds. Retrieved from emergingminds. com. au/resources/a ....

- Russell, T. G. (2007). Physical rehabilitation using telemedicine. *Journal of Telemedicine and Telecare*, *13*(5), 217-220.
- Sasikala, S., Indhira, K., & Chandrasekharan, V. M. (2018). Performance prediction of interactive telemedicine. *Informatics in Medicine Unlocked*, *30*, 1-8.
- Sharma, S., Ward, E. C., Burns, C., Theodoros, D., & Russell, T. (2011). Assessing swallowing disorders online: A pilot telerehabilitation study. *Telemedicine and E-Health*, 17(9), 688-695.
- Sharma, S., Ward, E. C., Burns, C., Theodoros, D., & Russell, T. (2013). Assessing dysphagia via telerehabilitation: Patient perceptions and satisfaction. *International Journal of Speech-Language Pathology*, 15(2), 176-183.
- Soldatova, L., Williams, C., Postma, G. N., Falk, G. W., & Mirza, N. (2020). Virtual dysphagia evaluation: practical guidelines for dysphagia management in the context of the COVID-19 pandemic. *Otolaryngology–Head and Neck Surgery*, 163(3), 455-458.
- Srivastava, A., Swaminathan, A., Chockalingam, M., Srinivasan, M. K., Surya, N., Ray, P., . . .
   Chitnis, S. (2021). Tele-neurorehabilitation during the COVID-19 pandemic: Implications for practice in low-and middle-income countries. *Frontiers in Neurology*, 12.
- Swales, M., Theodoros, D., Hill, A. J., & Russell, T. (2020). Speech-language pathologists' perceptions of the use of telepractice in the delivery of services to people with Parkinson's disease: A national pilot survey. *International journal of speech-language pathology*, 22(4), 387-398.
- Tinelli, F., Cioni, G., & Purpura, G. (2017). Development and implementation of a new telerehabilitation system for audiovisual stimulation training in hemianopia. *Frontiers in Neurology*, *8*, 621.
- Tucker, J. K. (2012). Perspectives of speech-language pathologists on the use of telepractice in schools: The qualitative view. *International Journal of Telerehabilitation*, 4(2), 67-72.
- Waite, M. C., Theodoros, D. G., Russell, T. G., & Cahill, L. M. (2010). Assessment of children's literacy via an Internet-based telehealth system. *Telemedicine and E-Health*, 16(5), 564-575.
- Wales, D., Skinner, L., & Hayman, M. (2017). The efficacy of telehealth-delivered speech and language intervention for primary school-age children: A systematic review. *International Journal of Telerehabilitation*, 9(1), 55.
- Ward, E., Crombie, J., Trickey, M., Hill, A., Theodoros, D., & Russell, T. (2009). Assessment of communication and swallowing post-laryngectomy: A telerehabilitation trial. *Journal of Telemedicine and Telecare*, 15(5), 232-237.
- Ward, E. C., & Burns, C. L. (2014). Dysphagia management via telerehabilitation: A review of the current evidence. *Journal of Gastroenterology and Hepatology Research*, *3*(5).
- Ward, E. C., Burns, C. L., Theodoros, D. G., & Russell, T. G. (2013). Evaluation of a clinical service model for dysphagia assessment via telerehabilitation. *International Journal of Telemedicine and Applications*, 2013.
- Ward, E. C., Burns, C. L., Theodoros, D. G., & Russell, T. G. (2014). Impact of dysphagia severity on clinical decision making via telerehabilitation. *Telemedicine and E-Health*, 20(4), 296-303.
- Ward, E. C., Sharma, S., Burns, C., Theodoros, D., & Russell, T. (2012). Managing patient factors in the assessment of swallowing via telerehabilitation. *International Journal of Telemedicine and Applications*, 2012.

- Ward, E. C., Wall, L. R., Burns, C. L., Cartmill, B., & Hill, A. J. (2017). Application of telepractice for head and neck cancer management: A review of speech language pathology service models. *Current opinion in otolaryngology and Head and Neck Surgery*, 25(3), 169-174.
- Weinstein, R. S., Lopez, A. M., Joseph, B. A., Erps, K. A., Holcomb, M., Barker, G. P., & Krupinski, E. A. (2014). Telemedicine, telehealth, and mobile health applications that work: Opportunities and barriers. *The American Journal of Medicine*, *127*(3), 183-187.
- WHO. (2018). Telehealth.
- WHO. (2021). Coronavirus disease (COVID-19) pandemic. *Geneva: World Health Organisation*.
- Wrape, E. R., & McGinn, M. M. (2019). Clinical and ethical considerations for delivering couple and family therapy via telehealth. *Journal of Marital and Family Therapy*, 45(2), 296-308.
- Wylie, K., McAllister, L., Davidson, B., & Marshall, J. (2013). Changing practice: Implications of the world report on disability for responding to communication disability in underserved populations. *International Journal of Speech-Language Pathology*, *15*(1), 1-13.

# Appendices

# Appendix A: Recruitment email example

I am Reshma Sreekumar, a master's student in the Speech and Language Therapy programme in the Institute of Education at Massey University, Auckland. My supervisors are Dr. Mershen Pillay and Dr. Sally Clendon.

I would like to ask for your assistance in recruiting participants for my master's research project. The aim of this project is to understand the perceptions of speech-language therapists in New Zealand on the use of telepractice with adults with dysphagia post-stroke and any factors they perceive as barriers and facilitators. The project involves completing an online survey.

To recruit participants, I would like to share the information sheet via the NZSTA social media pages. Is this something that the NZSTA would consider? I have attached my information sheet and low risk notification. Please let me know if there is any additional information I could provide.

Please do not hesitate to email on m.pillay@massey.ac.nz

or <u>Reshma.Sreekumar.1@uni.massey.ac.nz</u>, if you have any questions about this research project.

Here is the link to my survey

https://massey.au1.qualtrics.com/jfe/form/SV\_8rbVk7TUV8yTNeS

I look forward to hearing from you.

Kind regards, Reshma Sreekumar Master's student, Massey University, Auckland, New Zealand



## **Appendix B: Information Sheet**

Telepractice with Adults with Dysphagia Post-stroke: A Survey of New Zealand Speech-Language Therapists' Perceptions

## **Information Sheet**

## Invitation to participate in an online survey

I am Reshma Sreekumar, a Master's student in the Speech and Language Therapy programme at Massey University, Auckland. I would like to invite you to participate in a study exploring the perceptions of speech-language therapists (SLTs) with respect to telepractice with adults with dysphagia post-stroke.

The study is part of the requirements to fulfil my master's programme. It is being supervised by Dr. Mershen Pillay and Dr. Sally Clendon who are both Senior Lecturers in the Speech and Language Therapy programme in the Institute of Education at Massey University, Auckland. In this study, speech-language therapists in New Zealand who have experience engaging in telepractice with adults with dysphagia post-stroke will be invited to complete an anonymous online survey. I would like to invite you to participate. By participating in this study, you will be contributing to the growing body of information on the use of telepractice in dysphagia therapy.

## **Project Procedures**

The project will require completing an anonymous online survey at any time convenient to you. The survey will take approximately 20 minutes to complete. The survey responses will be analysed to summarise and explore the trends in this field.

Please note that by choosing to complete the online survey, your consent to participate in this study will be implied. The responses will remain anonymous and no identifying information will be collected. Once you have submitted your responses, it will not be possible to delete or amend your information due to the anonymous nature of the survey.

There are no anticipated risks involved with your participation. When the study is completed, the results will be presented in my master's thesis. The results may also be presented at scientific conferences and published in a research journal.

No publication arising out of the study shall have your name or personal details, as the survey is anonymous in nature.

Electronic data will be stored on password protected devices. The data will be stored for no longer than five years following the completion of this project. After that period, the University confidential waste service will be used to destroy any printed materials.

## Participant's rights

In following the ethical procedures, I assure you that you are under no obligation to consent to participate in this study. Once you decide to participate in this study, you have the right to:

- Decline to answer any question in the survey
- Contact the research team with any questions regarding the study

• Be given a summary of the findings when the project is completed – email your requests to Reshma.Sreekumar.1@uni.massey.ac.nz

## **Survey Link**

If you are interested in being involved in this project, you can complete the online survey by following this link https://massey.au1.qualtrics.com/jfe/form/SV\_8rbVk7TUV8yTNeS

Thank you for taking time to consider my request. If you have any questions or concerns regarding the proposed study, please contact one of the following researchers.

Project Contacts Reshma Sreekumar <u>Reshma.Sreekumar.1@uni.massey.a</u> c.nz

Dr. Mershen Pillay m.pillay@massey.ac.nz

## **Appendix C: Online questionnaire**

Telepractice with Adults with Dysphagia post-stroke: A Survey of New Zealand Speech-Language Therapists' Perceptions Questionnaire

## 1. Speech-Language Therapists' (SLTs) demographic information

1.1 How many years have you worked as a speech-language therapist?

1.2 What is your highest level of qualification in speech and language therapy?

- O Undergraduate level
- <sup>O</sup> Postgraduate Certificate
- <sup>O</sup> Postgraduate Diploma
- <sup>O</sup> Master's level
- O Doctoral level
- Other (please specify)

# 1.3 When in your education did you learn about dysphagia? (select all that apply)

- Undergraduate level
- Dostgraduate Certificate
- Postgraduate Diploma
- 🗌 Master's level
- Doctoral level
- Other (please specify)

## 1.4 What types of professional learning and development have you undertaken in telepractice? (select all that apply)

□ None

Undergraduate coursework

Postgraduate coursework

Peer learning

## 1.5 Where are you based in New Zealand?

- O Auckland
- O Northland
- O Waikato
- O Bay of Plenty
- O Gisborne
- O Hawke's Bay
- O Taranaki
- Wellington

# 1.6 What is your employment status?

- <sup>O</sup> Full time
- <sup>O</sup> Part time

- $\Box$  Conference, seminar or workshop
- Self-directed learning, e.g., reading books or journal articles
- Online learning, e.g., blogs, websites, participation in forums
- Other (please specify)
- O Whanganui-Manawatu
- O Malborough
- O Nelson
- O Tasman
- O West Coast
- Canterbury
- Otago
- Southland

# 1.7 Where do you provide clinical services for adults with dysphagia post-stroke? (select all that apply)

Public hospital

□ Private practice

 $\Box$  Aged care facility

- Other (please specify)
- 2. Clinical services for adults with dysphagia post-stroke

2.1 Approximately how many adults with dysphagia post-stroke have you worked with during the last two years?



2.2 What age group were these adults? (select all that apply)

- 20 to 39 years
- 40 to 60 years
- Above 60 years

# 3. Confidence with technology

3.1 Please rate your level of confidence using technology in general.

- <sup>O</sup> Poor
- <sup>O</sup> Fair
- <sup>O</sup> Good
- C Excellent

3.2 Please rate your level of confidence using technology in your clinical practice.

- <sup>O</sup> Poor
- <sup>O</sup> Fair
- <sup>O</sup> Good
- <sup>O</sup> Excellent
### 4. Telepractice services for adults with dysphagia post-stroke

- 4.1 What clinical services have you provided for adults with dysphagia post-stroke via telepractice? (select all that apply)
  - Assessment
  - Direct intervention (e.g., exercises)
  - Feeding trials
  - Rehabilitation/compensatory strategies (e.g., muscle strength training)
  - Education for clients and caregivers
  - Other (please specify)

4.2 What platforms have you used when providing clinical services for adults with dysphagia post-stroke via telepractice? (select all that apply)

- Audio calls (phone, skype)
- <sup>□</sup> Video-conferencing
- Other (please specify)

4.3 Do you use any of the following to augment your clinical swallow evaluation via telepractice? (select all that apply)

- Cervical Auscultation
- Dulse Oximetry
- SEMG- Surface Electromyography
- Other (please specify)
- $\Box$  None of the above

4.4 What client-related factors do you consider when deciding to offer adults with dysphagia post-stroke clinical services via telepractice? (select

# all that apply)

- Cognitive/behavioral ability (e.g., cognitive functioning, multitasking, attention)
- Physical/sensory ability (e.g., hearing, vision, manual dexterity, sitting tolerance)
- Communication ability (e.g., auditory comprehension, literacy, speech intelligibility)
- Technology skills

- C Access to hardware (e.g., laptop, iPad)
- CACCESS to software (e.g., skype, zoom)
- Internet qaulity
- Access to support pers<u>on (e.g.,</u> caregiver)
- Cther (please specify)

# 5. Expansion of clinical services for adults with dysphagia post-stroke via telepractice

- 5.1 Do you plan to expand your clinical services for adults with dysphagia post-stroke via telepractice?
  - <sup>O</sup> Yes
  - <sup>O</sup> No

# 5.2 How do you plan to expand your clinical services via telepractice? (select all that apply)

- Increase number of clients
- Expand range of equipment
- Expand range of services
- □ Other (please specify)
- 6. Benefits of clinical services for adults with dysphagia post-stroke via telepractice

### 6.1 What are the client-related benefits?



6.2 What are the clinician-related benefits?



6.3 What are the workplace/organisation-related benefits?



7. Facilitators of clinical services for adults with dysphagia post-stroke via telepractice

7.1 What client-related facilitators have you experienced (e.g., technology skills, internet quality)?



7.2 What clinician-related facilitators have you experienced (e.g., experience, confidence)?



7.3 What organization/workplace related facilitators have you experienced (e.g., quite space, IT support)?



#### 8. Barriers to clinical services for adults with dysphagia post-stroke via telepractice



# 8.1 What client-related barriers have you experienced?

#### 9. Beliefs about telepractice

9.1 Indicate your level of agreement with the following statements:

|   | Strongly disagree   | Somewhat disagree  | Neither agree nor disagree  | Somewhat agree  | Strongly agree   |
|---|---|--|---|---|--|
| Telepractice is an<br>appropriate method to<br>deliver clinical<br>services for adults with<br>dysphagia post-stroke. | <ul> <li>Telepractice is an appropriate method to deliver clinical services for adults with dysphagia post-stroke.</li> </ul> | <sup>O</sup> Telepractice is an<br>appropriate method to<br>deliver clinical<br>services for adults with<br>dysphagia post-stroke. | • Telepractice is an appropriate method to deliver clinical services for adults with dysphagia post-stroke. | • Telepractice is an appropriate method to deliver clinical services for adults with dysphagia post-stroke. | • Telepractice is an appropriate method to deliver clinical aservices for adults with dysphagia post-stroke. |

|   | Strongly disagree<br>Strongly disagree  | Somewhat disagree  | Neither agree nor<br>disagree<br>Neither agree nor   | Somewhat agree  | Strongly agree  |
|---|---|--|--|---|---|
| will<br>ecome a<br>delivery<br>peech-<br>rapy<br>adults with<br>ost-stroke. | C Telepractice will<br>eventually become a<br>mainstream delivery<br>method for speech-<br>language therapy<br>services for adults withs<br>dysphagia post-stroke. O<br>Strongly disagree | C Telepractice will<br>eventually become a<br>mainstream delivery<br>method for speech-<br>language therapy<br>services for adults with<br>dysphagia post-stroke.<br>Somewhat disagree | <ul> <li>C Telepractice will<br/>eventually become a<br/>mainstream delivery<br/>method for speech-<br/>language therapy<br/>services for adults with<br/>dysphagia post-stroke.<br/>Neither agree nor<br/>disagree</li> </ul> | C Telepractice will<br>eventually become a<br>mainstream delivery<br>method for speech-<br>language therapy<br>services for adults with<br>dysphagia post-stroke.<br>Somewhat agree | C Telepractice will<br>eventually become a<br>mainstream delivery<br>method for speech-<br>language therapy<br>services for adults with<br>dysphagia post-stroke.<br>Strongly agree |
| most<br>w Zealand<br>gia post-<br>ave access<br>rvices via                  | In 10 years, most<br>adults in New Zealand a<br>with dysphagia post-<br>stroke will have access a<br>to clinical services via<br>telepractice. Strongly to<br>disagree                    | In 10 years, most<br>adults in New Zealand<br>with dysphagia post-<br>stroke will have access<br>to clinical services via<br>elepractice. Somewhat<br>disagree                         | <sup>O</sup> In 10 years, most<br>adults in New Zealand<br>with dysphagia post-<br>stroke will have access<br>to clinical services via<br>telepractice. Neither<br>agree nor disagree  | • In 10 years, most<br>adults in New Zealand<br>with dysphagia post-<br>stroke will have access<br>to clinical services via<br>telepractice. Somewhat<br>agree                      | • In 10 years, most<br>adults in New Zealand<br>with dysphagia post-<br>stroke will have access<br>to clinical services via<br>telepractice. Strongly<br>agree                      |
| at 10 years,<br>' limited<br>nd interest<br>y will<br>of a<br>epractice.    | • Over the next 10<br>years, older clients'<br>limited familiarity and<br>interest in technology<br>will become less of a<br>barrier to telepractice.                                     | Over the next 10<br>years, older clients'<br>limited familiarity and<br>interest in technology<br>will become less of a<br>barrier to telepractice.                                    | Over the next 10<br>years, older clients'<br>limited familiarity and<br>interest in technology<br>will become less of a<br>barrier to telepractice.  | • Over the next 10<br>years, older clients'<br>limited familiarity and<br>interest in technology<br>will become less of a<br>barrier to telepractice.                               | Over the next 10<br>years, older clients'<br>limited familiarity and<br>interest in technology<br>will become less of a<br>barrier to telepractice.                                 |

Telepractice will eventually become a mainstream delivery method for speechlanguage therapy services for adults with dysphagia post-stroke.

In 10 years, most adults in New Zealand with dysphagia poststroke will have access to clinical services via telepractice.

Over the next 10 years, older clients' limited familiarity and interest in technology will become less of a barrier to telepractice.

100

|   | Strongly disagree   | Somewhat disagree   | Neither agree nor<br>disagree  | Somewhat agree   | Strongly agree  |
|---|---|---|--|--|---|
|   | Strongly disagree   | Somewhat disagree   | Neither agree nor<br>disagree  | Somewhat agree   | Strongly agree  |
| I will only use<br>telepractice if in-<br>person delivery is not<br>an option.  | ○ I will only use<br>telepractice if in-<br>person delivery is not<br>an option. Strongly<br>disagree   | C I will only use<br>telepractice if in-<br>person delivery is not<br>an option. Somewhat<br>disagree   | C I will only use<br>telepractice if in-<br>person delivery is not<br>an option. Neither<br>agree nor disagree   | C I will only use<br>telepractice if in-<br>person delivery is not<br>an option. Somewhat<br>agree   | C I will only use<br>telepractice if in-<br>person delivery is not<br>an option. Strongly<br>agree                                      |
| You cannot develop<br>the same level of<br>client-clinician rapport<br>via telepractice as is<br>achieved in-person.                                | • You cannot<br>develop the same level<br>of client-clinician<br>rapport via telepractice<br>as is achieved in-<br>person. Strongly<br>disagree | • You cannot<br>develop the same level<br>of client-clinician<br>rapport via telepractice<br>as is achieved in-<br>person. Somewhat<br>disagree | • You cannot<br>develop the same level<br>of client-clinician<br>rapport via telepractice<br>as is achieved in-<br>person. Neither agree<br>nor disagree | • You cannot<br>develop the same level<br>of client-clinician<br>rapport via telepractice<br>as is achieved in-<br>person. Somewhat<br>agree | You cannot<br>develop the same level<br>of client-clinician<br>rapport via telepractice<br>as is achieved in-<br>person. Strongly agree |
| My<br>workplace/organisation<br>is open and interested<br>in using telepractice to<br>provide services for<br>adults with dysphagia<br>post-stroke. | <sup>O</sup> My<br>workplace/organisation   | • My<br>workplace/organisation  | • My<br>workplace/organisation   | • My<br>workplace/organisation   | • My<br>workplace/organisation  |
|   | is open and interested<br>in using telepractice to<br>provide services for<br>adults with dysphagia<br>post-stroke. Strongly<br>disagree        | is open and interested<br>in using telepractice to<br>provide services for<br>adults with dysphagia<br>post-stroke. Somewhat<br>disagree        | is open and interested<br>in using telepractice to<br>provide services for<br>adults with dysphagia<br>post-stroke. Neither<br>agree nor disagree        | is open and interested<br>in using telepractice to<br>provide services for<br>adults with dysphagia<br>post-stroke. Somewhat<br>agree        | is open and interested<br>in using telepractice to<br>provide services for<br>adults with dysphagia<br>post-stroke. Strongly<br>agree   |
| Telepractice can be   | • Telepractice can be   | <sup>O</sup> Telepractice can be  | <sup>O</sup> Telepractice can be   | • Telepractice can be  | • Telepractice can be   |

| used to provide a<br>service that is as<br>effective as in-person<br>delivery.   | Strongly disagree<br>used to provide a<br>service that is as<br>effective as in-person<br>delivery. Strongly   | Somewhat disagree<br>used to provide a<br>service that is as<br>effective as in-person<br>delivery. Somewhat  | Neither agree nor<br>disagree<br>used to provide a<br>service that is as<br>effective as in-person<br>delivery. Neither agree   | Somewhat agree<br>used to provide a<br>service that is as<br>effective as in-person<br>delivery. Somewhat  | Strongly agree<br>used to provide a<br>service that is as<br>effective as in-person<br>delivery. Strongly  |
|--|--|---|---|--|--|
| Telepractice will only<br>benefit adults with<br>dysphagia post-stroke<br>who live in regional<br>towns or rural areas,<br>not city centres. | <ul> <li>Telepractice will<br/>only benefit adults<br/>with dysphagia post-<br/>stroke who live in<br/>regional towns or rural<br/>areas, not city centres.<br/>Strongly disagree</li> </ul> | Consider the second stroke who live in regional towns or rural areas, not city centres. Somewhat disagree   | <ul> <li>Telepractice will<br/>only benefit adults<br/>with dysphagia post-<br/>stroke who live in<br/>regional towns or rural<br/>areas, not city centres.<br/>Neither agree nor<br/>disagree</li> </ul> | Telepractice will<br>only benefit adults<br>with dysphagia post-<br>stroke who live in<br>regional towns or rural<br>areas, not city centres.<br>Somewhat agree    | agree<br>Telepractice will<br>only benefit adults<br>with dysphagia post-<br>stroke who live in<br>regional towns or rural<br>areas, not city centres.<br>Strongly agree |
| Telepractice has the<br>potential to improve<br>the level of care I am<br>currently providing for<br>adults with dysphagia<br>post-stroke.   | C Telepractice has<br>the potential to<br>improve the level of<br>care I am currently<br>providing for adults<br>with dysphagia post-<br>stroke. Strongly<br>disagree                        | C Telepractice has<br>the potential to<br>improve the level of<br>care I am currently<br>providing for adults<br>with dysphagia post-<br>stroke. Somewhat<br>disagree | • Telepractice has<br>the potential to<br>improve the level of<br>care I am currently<br>providing for adults<br>with dysphagia post-<br>stroke. Neither agree<br>nor disagree                            | C Telepractice has<br>the potential to<br>improve the level of<br>care I am currently<br>providing for adults<br>with dysphagia post-<br>stroke. Somewhat<br>agree | C Telepractice has<br>the potential to<br>improve the level of<br>care I am currently<br>providing for adults<br>with dysphagia post-<br>stroke. Strongly agree          |

# **10. Additional comments**

10.1 Do you have any additional comments related to working with adults with dysphagia post-stroke via telepractice?



#### **APPENDIX D: Ethics approval letter**

Ethics Notification Number: 4000024328

Title: Telepractice with Adults with Dysphagia post-stroke: A Survey of New Zealand Speech-Language Therapists' Perceptions

Thank you for your notification which you have assessed as Low Risk.

Your project has been recorded in our system which is reported in the Annual Report of the Massey University Human Ethics Committee.

The low-risk notification for this project is valid for a maximum of three years. Please note that travel undertaken by students must be approved by the supervisor and the relevant Pro Vice-Chancellor and be in accordance with the Policy and Procedures for Course-Related Student Travel Overseas. In addition, the supervisor must advise the University's Insurance Officer.

A reminder to include the following statement on all public documents:

"This project has been evaluated by peer review and judged to be low risk. Consequently, it has not been reviewed by one of the University's Human Ethics Committees. The researcher(s) named in this document are responsible for the ethical conduct of this research. If you have any concerns about the conduct of this research that you want to raise with someone other than the researcher(s), please contact Professor Craig Johnson, Director (Research Ethics), email: humanethics@massey.ac.nz. "

Please note that if a sponsoring organisation, funding authority or a journal in which you wish to publish required evidence of committee approval (with an approval number), you will have to complete the application form again answering yes to the publication question to provide more information to go before one of the University's Human Ethics Committees. You should also note that such an approval can only be provided prior to the commencement of

104

the research.

You are reminded that staff researchers and supervisors are fully responsible for ensuring that the information in the low risk notification has met the requirements and guidelines for submission of a low-risk notification.

If you wish to print an official copy of this letter, please login to the RIMS system, and under the Reporting section, View Reports you will find a link to run the LR Report.

Yours sincerely Professor Craig Johnson Chair, Human Ethics Chairs' Committee and

Director (Research Ethics)